



WORKING PAPERS IN RESPONSIBLE BANKING & FINANCE

Audit Committee Oversight and Bank Financial Reporting Quality

By Dimitris K. Chronopoulos, Lemonia M. Rempoutsika, John O.S. Wilson

Abstract: This study investigates the impact of audit committee oversight on the financial reporting quality of US bank holding companies. To overcome identification concerns, we use Section 165h of the Dodd Frank Wall Street Reform and Consumer Protection Act, which requires publicly traded bank holding companies with assets exceeding \$10 billion to have separate audit and risk committees. We utilize a difference-in-differences framework where our treatment group comprises bank holding companies that were required to separate audit and risk oversight functions following the introduction of Section 165h, and our control group comprises counterparts that already had separate audit and risk committees prior to the passage of Section 165h. We estimate the difference in the behaviour of treated bank holding companies between the pre- and post-implementation period of Section 165h with the same difference in the behaviour of control group counterparts and find that the separation of audit and risk committees leads to an improvement in financial reporting quality. We attribute the observed improvements in financial reporting quality to the increased focus of audit committees arising from a reduction in the volume and complexity of tasks undertaken following the implementation of Section 165h.

WP Nº 21-015

4th Quarter 2021



University of St Andrews Scotland's first university

ALT IN THE REAL

600 YEARS 1413 – 2013

Audit Committee Oversight and Bank Financial Reporting Quality

Dimitris K. Chronopoulos, Lemonia M. Rempoutsika, John O.S. Wilson

Centre for Responsible Banking & Finance School of Management, University of St Andrews

Abstract

This study investigates the impact of audit committee oversight on the financial reporting quality of US bank holding companies. To overcome identification concerns, we use Section 165h of the Dodd Frank Wall Street Reform and Consumer Protection Act, which requires publicly traded bank holding companies with assets exceeding \$10 billion to have separate audit and risk committees. We utilize a difference-in-differences framework where our treatment group comprises bank holding companies that were required to separate audit and risk oversight functions following the introduction of Section 165h, and our control group comprises counterparts that already had separate audit and risk committees prior to the passage of Section 165h. We estimate the difference in the behaviour of treated bank holding companies between the pre- and post-implementation period of Section 165h with the same difference in the behaviour of control group counterparts and find that the separation of audit and risk committees leads to an improvement in financial reporting quality. We attribute the observed improvements in financial reporting quality to the increased focus of audit committees arising from a reduction in the volume and complexity of tasks undertaken following the implementation of Section 165h.

JEL Classification: G21; G28; M41; M48

Keywords: Audit committee; Financial reporting quality; Bank holding companies; Dodd-Frank Act; Loan loss provisioning; Board busyness.

1. Introduction

A lack of internal controls and audit committee oversight contributed to the management failures and subsequent financial instability at many banks during the global financial crisis (Kirkpatrick, 2009; Financial Crisis Inquiry Commission, 2011; Hagendorff, 2019). As part of the post crisis reform agenda, new regulations were enacted, which require banks to engage in more effective audit committee oversight. In this study, we investigate the impact of audit committee oversight on the financial reporting quality of US bank holding companies.

Improvements in the quality of financial reporting information can reduce information asymmetries and improve the ability of regulators, auditors, and other outside stakeholders in monitoring the performance and risk of firms (Bushman and Smith, 2001). However, assessing the financial reporting quality of banks presents a significant challenge given that assets held by financial institutions are often complex, opaque and difficult to value (Morgan, 2002; Jones et al., 2012; Flannery et al., 2013; Jiang et al., 2016). Given the critical role that banks play in the financial system and real economy, it is crucial that disclosed financial information is accurate and transparent so that outside stakeholders (depositors, equity-holders, bondholders, regulators) can accurately assess and monitor the financial condition of banks.¹

In order to investigate the impact of audit committee oversight on the financial reporting quality of bank holding companies, we use the US banking industry as a setting. Prior to the financial crisis, the audit committees of bank holding companies were tasked with the oversight and monitoring of both financial reporting quality and risk management. However, following the global financial crisis of 2007-2009, policymakers strengthened prudential standards for corporate governance arrangements at banks via the passing of the Dodd-Frank Wall Street Reform and Consumer Protection Act of 2010. Section 165h of the Dodd Frank Act requires that publicly traded bank holding companies with total assets exceeding \$10 billion (as of 30th June 2014) establish a risk committee that is distinct and independent of the existing audit committee. The deadline for compliance was 1st July 2015 (Federal Register, 2014).

Prior to the passage of Section165h of the Dodd-Frank Act, in addition to the core task of overseeing financial reporting and effective communication to stakeholders, audit

¹ Banks operate the payments system, act as a conduit for monetary policy and are a major source of credit for households, corporations and governments (Allen et al., 2019). They act as a haven for household, corporate and government deposits funds and create liquidity by transforming relatively liquid liabilities in relatively illiquid assets both on- and off- balance sheet (Berger et al., 2020).

committees were required to identify, monitor and report on significant risk exposures, and articulate any actions taken to address these. These arrangements led to possible concerns that members of bank audit committees had insufficient capacity to handle the volume and multiplicity of tasks required to ensure appropriate financial reporting quality (Ernst & Young, 2014; KPMG, 2014).

Regulatory reforms proposed under Section 165h of the Dodd-Frank Act provide the setting for the current study, which investigates the impact of audit committees on the financial reporting quality of bank holding companies. Given that Section 165h requires a separation of audit and risk committees, we posit that following the introduction of the new regulation, audit committees have more capacity to focus on core internal control and audit functions. This increased focus leads to an improvement in financial reporting quality.

The setting used for the current study (which allows us to identify bank holding companies subject to the introduction of Section 165h versus counterparts that were not subject to the new regulation) allows for a more rigorous research design than that utilized by many prior studies of financial reporting quality (Carcello and Neal 2000; Klein 2002; Abbott et al., 2003, 2004; Bédard et al., 2004; Karamanou and Vafeas, 2005). While all publicly traded banks with total assets exceeding \$10 billion were required to comply with Section 165h, some bank holding companies already had distinct audit and risk committees prior to the introduction of the new regulation. Consequently, it is possible to identify a group of affected and unaffected bank holding companies before (pre-treatment) and after (post-treatment) the introduction of Section 165h. This provides the basis for a robust research design to test our research hypothesis.

Financial reporting quality cannot be observed directly. Consequently, following established practice in prior literature, we utilize discretionary loan loss provisions as a proxy for financial reporting quality. Loan loss provisions are the most important accrual on bank balance sheets and should reflect the quality of loan portfolios. However, prior evidence suggests that bank managers exercise considerable discretion in loan loss provisioning via earnings smoothing, signaling and capital management activities (Beatty and Liao, 2014).²

² Schipper and Vincent (2003), Francis et al. (2006) and Dechow et al. (2010) provide extensive overviews of the measurement and determinants of financial reporting quality. Banks may have an incentive to smooth reported earnings via loan loss provisions to make reported earnings appear stable over time and meet pre-defined prudential regulatory requirements or satisfy opportunistic financial reporting objectives. The inclusion of loan loss reserves in the calculation of regulatory capital can lead bank managers to manipulate loan loss provisions in order to report regulatory capital above a certain minimum. Banks may also use loan loss provisions to signal information regarding loan quality and future earnings prospects to firm stakeholders (investors, customers, regulators).

Following prior literature, we use the absolute value of residuals derived from an estimable model, which allows us to disentangle the discretionary and non-discretionary components of loan loss provisions. The resultant discretionary loan loss provisions are used as our primary outcome variable of interest in our empirical analysis.

In order to assess the impact of the separation of audit and risk committee (via the introduction of Section 165h of the Dodd-Frank Act) on financial reporting quality, we adopt a difference-in-differences framework. Our treatment group comprises bank holding companies that were required to separate audit and risk oversight functions following the introduction of Section 165h. Our control group comprises bank holding companies that already had separate audit and risk committees prior to the passage of Section 165h, and thus were unaffected by the regulatory change. The period of our investigation spans 2007 through 2016, and straddles the enactment of Section 165h of the Dodd-Frank Act.

By way of preview, the results of an extensive empirical analysis suggest that bank holding companies affected by the separation of audit and risk committees reduce discretionary loan loss provisions relative to unaffected counterparts. Our results are economically significant and indicate that the separation of the audit and risk committee leads to an improvement in financial reporting quality. We conduct an additional analysis to investigate the mechanism through which the separation of audit and risk committees improves financial reporting quality. Specifically, we examine whether several characteristics related to audit committee effectiveness affect bank financial reporting quality following the introduction of Section 165h of Dodd-Frank Act. The results suggest that bank holding companies with busy audit committee members serving on other committees within the same (*intra-bank busyness*) or other bank boards (*inter-bank busyness*) experience a decline in financial reporting quality relative to counterparts with less busy directors. We rule out an alternative explanation for our results that the over-provisioning for loan losses taking place prior to Section 165h was a prudential response of audit committee members to higher credit risk facing treated banks.

Our results are robust to alternative measures of financial reporting quality and are not affected by other events such as: the participation of bank holding companies in the Trouble Asset Relief Program; changes in state corporate income taxes facing banks; or participation in stress testing programs, which occurred around the time of the introduction of Section 165h of the Dodd-Frank Act. We assess the internal validity of our findings via a placebo test where we assume falsely that the separation of the audit and risk committees took place in 2008 rather than 2010. The results of this placebo test are statistically insignificant, thus supporting the causal interpretation of our main findings.

Our study contributes to four strands of literature. First, we add to literature that assesses the impact of audit committees and regulatory interventions on bank financial reporting quality.³ Cornett et al. (2009) provide evidence that effective audit committees constrain earnings management behavior at large US bank holding companies prior to the passage of the Sarbanes-Oxley Act. Altamuro and Beatty (2010) assess the impact of the introduction and implementation of internal control requirements (as part of the provisions of the Federal Deposit Insurance Corporation Improvement Act) on the financial reporting quality of banks.⁴ The authors find that the implementation of mandated internal control requirements leads to improved financial reporting quality. Delis et al. (2018) find that US banks subject to accounting related regulatory enforcement actions subsequently improved financial reporting quality. In the current study, we augment this literature to investigate the impact of regulations regarding the scope of audit committees on financial reporting quality. We find that the separation of audit and risk committees mandated by the introduction of Section 165h of the Dodd-Frank Act leads to an improvement in the financial reporting quality of bank holding companies.

Second, we contribute to an emerging literature that examines whether the increasing scope and multiplicity of tasks required of audit committees leads members to become overloaded with responsibilities, and thus less able to execute core functions of overseeing financial reporting quality. Prior research shows that firms with a higher proportion of audit committee members who hold fewer multiple directorships (are less inter-bank busy) or sit on fewer committees on the same board (are less intra-bank busy) produce higher quality financial reporting information (Fich and Shivdasani, 2006 Dhaliwal et al., 2010; Sun and Liu, 2014; Chen and Wu, 2016). Using a sample of non-financial firms, Ashraf et al., (2020) investigate whether increases in audit committee responsibilities impair the quality and reliability of financial statements. The authors find that audit committees, which are allocated responsibilities (such as risk management) unrelated to financial reporting produce lower quality financial information. We complement this literature using a sample of US bank

³ Extensive overviews of the financial reporting quality literature in banking are provided by Wall and Koch (2000), Beatty and Liao (2014) and Bushman (2014). A selective overview of this literature is also provided in section 3 below.

⁴ Banks with total assets exceeding \$500 million were required to compile a report (certified by the external auditor) which disclosed the effectiveness of internal controls over financial reporting. LaFond and You (2010) suggest that the findings produced by Altamuro and Beatty (2010) are confounded by other events taking place during the long sample period.

holding companies. The difference-in-differences design allows us to identify the causal impact of the separation of audit and risk committees on financial reporting quality. We find that affected bank holding companies improve reporting quality (by constraining earnings management) via a reduction in discretionary loan loss provisions. This finding is robust to alternative measures of financial reporting quality such as small positive earnings changes and avoiding (negative earnings surprises).

Third, our study contributes to the literature on bank opacity. The increasing size and complexity of financial institutions has gained the attention of both academics and policymakers following the global financial crisis (Copeland, 2012; Avraham et al., 2012; Cetorelli et al, 2014; Liu et al, 2020). Cetorelli and Goldberg (2014) suggest that managerial incentives are a significant contributory factor in the increasing complexity and resultant opacity of banks. Our results suggest that by allowing audit committees to focus on core functions, financial reporting quality and resultant bank transparency are improved. This improves the quality of information available to bank stakeholders. As such the results of our analysis have relevance for market discipline in the banking industry given that transparent, timely and reliable information can aid internal (depositors) and external stakeholders (investors, regulators) in taking actions to mitigate excessive risk (Bushman and Williams, 2012; Nguyen, 2013; Cohen et al. 2014; Flannery, 2009; Flannery and Bliss, 2019).

Finally, we contribute to an emerging literature regarding the impact of various provisions of the Dodd-Frank Act on bank behavior (FSOC, 2011; FSOC, 2012; FSOC; 2016). Prior evidence suggests that the implementation of various provisions embodied in the Dodd-Frank Act improved financial stability and market discipline. Balasubramnian and Cyree (2014), Akhigbe et al. (2016) and Gao et al. (2018) provide evidence which suggests that the Dodd-Frank Act lessened too-big-to-fail (TBTF) issues at large banks. Evans and Schwartz (2014) offer evidence that the burden of regulatory compliance is disproportionately higher for smaller banks, while Cyree (2016) finds that small bank compliance costs increase significantly. Leledakis and Pyrgiotakis (2019) find that the acquisition of small banks increased after the passage of the Dodd-Frank Act. Our results suggest that imposing stricter requirements on large bank holding companies leads to changes in financial reporting quality.

The rest of this paper is organized as follows. Section 2 provides a background to the functions and evolution of audit committees. In Section 3 we provide a review of the research evidence on bank financial reporting quality. This section also reviews research that

investigates the role of audit committees in influencing the behavior of both non-financial and financial firms. Section 4 discusses the research design. In section 5, we discuss the dataset used, while section 6 presents the results of our empirical analysis. Section 7 presents the results for a series of additional tests, while Section 8 concludes.

2. Background

The primary purpose of an audit committee is to oversee the financial reporting process, hire external auditors and evaluate the results produced by internal and external audit. Members of the audit committee review significant financial reporting issues and make judgements, which affect the preparation of financial statements, interim reports, preliminary announcements, and other formal statements. The audit committee also reviews the clarity and completeness of any disclosures made in financial statements. It is also the responsibility of the audit committee to monitor and review the internal audit activities. In cases where there is no internal audit function, the audit committee considers (on an annual basis) whether there is a need for an internal audit function and reports the outcome of these deliberations to the board. A relevant section of the annual report is used to explain any absence of internal audit. In the absence of separate arrangements, the audit committee also assesses the scope and effectiveness of the systems established by management to identify, manage, and monitor financial and non-financial risks. Members of the audit committee are also responsible for addressing any issues arising from risk management procedures and report them to the board. Finally, the audit committee is responsible for overseeing relations with the external auditor. Members of the audit committee have primary responsibility for making recommendations on the appointment, reappointment, and removal of external auditors. These recommendations are made to the board, and then put to a shareholder vote at the annual general meeting. If the board does not accept the audit committee's recommendation, it should provide a rationale for any contrary position in the annual report.

The origin of audit committees in the United States dates back to 1940, when the US Securities and Exchanges Commission (SEC) recommended that listed (financial and non-financial) firms appoint an audit committee to propose and subsequently monitor the work of an external auditor. In the late 1980s, the Treadway Commission strengthened the supervisory and oversight roles of audit committees in the financial reporting process.⁵ This

⁵ The Treadway Commission was formed in 1985 to investigate, analyze and make recommendations regarding fraudulent corporate financial reporting. Audit committees were identified by the commission as essential parts of any system designed to prevent fraudulent financial reporting (Sommer, 1991).

continued throughout the 1990s as relevant authorities published guidelines aimed at improving and enforcing audit committee quality and performance (Blue Ribbon Committee, 1999; Securities and Exchanges Commission, 1999b; American Institute of Certified Public Accountants, 1999; National Association of Corporate Directors, 2000). While the aforementioned changes strengthened the supervisory and oversight roles of audit committees, high profile corporate failures (including Enron and WorldCom) raised further concerns regarding the quality of audit committees and corporate governance more generally. This led to an overhaul and implementation of a variety of regulations regarding audit committee effectiveness, many of which were included in the Sarbanes-Oxley Act (Healy and Palepu, 2003; Coffee, 2005).⁶

Passed in 2002, the Sarbanes-Oxley Act established standards relating to the audit committees of publicly listed firms. Sections 301 and 407 of the Act significantly increased the importance and responsibilities of audit committees (Linck et al., 2009; Beasley et al. 2009; Cohen et al., 2010). Under the provisions of Sarbanes Oxley Act, audit committees were deemed responsible for monitoring financial reporting processes, evaluating internal control and organizing and overseeing the external audit. Moreover, audit committees should be composed of independent members including at least one financial expert.

In 2002, the New York Stock Exchange (NYSE) Corporate Accountability and Listing Standards Committee recommended that audit committees should discuss guidelines and policies governing the assessment and management of risk (Bates and Leclerc, 2009). Critics of changes enacted under the Sarbanes-Oxley Act and the earlier Blue-Ribbon Committee expressed concerns that the scale and scope of activities undertaken by audit committees had expanded too far, leading to them becoming over-stretched and deviating from their core function and area of expertise (Zaman, 2001; Claypool et al., 2004).

In the wake of the global financial crisis, policy makers questioned the appropriateness of the corporate governance regimes in the banking industry. Risk management and corporate governance deficiencies were at the core of many bank failures.⁷ As a result, the role of audit committees attracted scrutiny (Aebi et al., 2012). A KPMG survey suggests that members of the audit committee expressed concerns that current

⁶ Zalewska (2014) provides a review of corporate governance reforms in the US and the UK as a result of major accounting scandals.

⁷ The weakness observed in the banking sector during the financial crisis was at least partly due to an accumulation of excessive risks by some banks (Srivastav and Hagendorff, 2016). This raised questions that directors who served bank boards were ineffective monitors of risk (Kirkpatrick, 2009; Hagendorff, 2019).

responsibilities for risk oversight distracted the audit committee from the core function of monitoring financial reporting quality (KMPG, 2009).

Based on the premise that the financial crisis was caused, at least in part, by corporate governance failures, the US Senate introduced, but failed to pass the Shareholder Bill of Rights Act, which recommended assigning board risk oversight responsibility to a separate risk committee (Honig, 2009). In July 2010, the Dodd-Frank Wall Street Reform and Consumer Protection Act of 2010 was passed. The Act included a specific provision, Section 165h, which required publicly traded bank holding companies with total assets exceeding \$10 billion to establish a stand-alone risk committee independent from the existing audit committee. Proponents of this policy argued that the establishment of stand-alone risk committees would bring relief to over-stretched audit committees by allowing them to focus on their core functions (Protiviti, 2011).

3. Literature Review

In this section, we present a selective review of the relevant literature. Section 3.1 provides an overview of the financial reporting quality literature. In Section 3.2 and 3.3 we provide an overview of the literature regarding audit committee effectiveness for non-financial and financial services firms respectively.

3.1 Financial reporting quality in the banking industry

The importance of bank financial reporting quality for the banking industry and global economy has been stressed by bank regulators for many years (Crockett, 2002). Financial reporting information that is transparent, timely and reliable can assist stakeholders in assessing the financial condition of banks (Flannery, 2009; Flannery and Bliss, 2019). Consequently, it is critical to understand the underlying factors driving the financial reporting quality of banks.

In the empirical bank accounting literature, financial reporting quality is measured most commonly by loan loss provisions. Evidence produced by early studies suggests that the accuracy of bank loan loss provisioning plays a critical role in determining the quality of financial reporting information (Wall and Koch, 2000).⁸ However, there is a general consensus that bank managers routinely use discretion in loan loss provisioning in such a

⁸ Loan loss provisions are made up of non-discretionary and discretionary components. The non-discretionary component represents loan loss provisions made to cover expected credit losses (Wahlen, 1994; Beaver and Engel, 1996). This component is related to the identification of non-performing loans, and exhibits a cyclical pattern (decreasing during periods of economic buoyancy and increasing as economic conditions deteriorate (Laeven and Majnoni, 2003; Bikker and Metzemakers, 2005). The discretionary component captures loan loss provisions made for managerial objectives such as income smoothing, capital management or signalling (Ahmed et al., 1999; Kanagaretnam et al., 2004; Fonseca and Gonzalez, 2008; Kanagaretnam et al., 2009).

way as to make reporting earnings appear stable over time (Greenawalt and Sinkey, 1988; Wahlen, 1994).

A number of empirical studies examine the relationship between discretionary loan loss provisions and earnings. No clear consensus emerges with some studies finding a positive correlation between earnings and discretionary loan loss provisions, consistent with earnings smoothing (Collins et al., 1995; Kanagaretnam et al., 2004; Liu and Ryan, 2006; El Sood, 2012; Bouvatier et al., 2014; Curcio and Hasan, 2015; Curcio et al., 2017), while others fail to find any significant association between earnings and loan loss provisions (Beatty et al., 1995; Ahmed et al., 1999).

Another strand of literature investigates whether banks use discretionary loan loss provisioning in order to manage capital. Bank managers have incentives to influence the level of reported loan loss provisions in order to meet minimum capital requirements (Moyer, 1990; Barth et al., 2017). Prior studies that test the capital management hypothesis focus on the association between discretionary loan loss provisions and Tier 1 capital before loan loss provisions. Much of the evidence presented (with the notable exception of Beatty et al., 1995 who find that discretionary loan loss provisions are positively related to capital) suggests that there is a negative correlation between discretionary loan loss provisions and Tier 1 capital before deductions for loan losses. This is consistent with the view that bank use loan losses to manage capital (Beaver and Engel, 1996; Collins et al., 1995; Kim and Kross, 1998; Ahmed et al., 1999; and Curcio and Hasan, 2015).

Prior research also investigates whether bank managers use loan loss provisions to signal private information regarding bank loan portfolio quality to outsiders (Wahlen, 1994; Liu and Ryan, 1995; Beaver and Engel, 1996). The signaling hypothesis derives from the positive association between discretionary loan loss provisions and one-year forward earnings. Kanagaretnam et al. (2003) find that managers of banks with low external credit ratings use loan loss provisions to increase the level of reported earnings in order to send a positive signal regarding future earnings prospects. In a similar vein, Kanagaretnam et al. (2005) produce evidence, which suggests that banks use discretionary loan loss provision to signal future earnings prospects.

An important strand of the banking literature examines how banks vary loan loss provisions over the business cycle. Much of the evidence presented suggests that loan loss provisions are pro-cyclical with banks decreasing (increasing) lending and increasing (decreasing) loan loss provisions during economic downturns (upturns). This amplifies general economic and financial conditions (Laeven and Majnoni, 2003; Bikker and Metzemakers, 2005; Bouvatier and Lepetit, 2008; Beatty and Liao, 2011; Agenor and Zilberman, 2015; Huizinga and Laeven, 2019).

3.2 The Role of Audit Committees in non-financial firms

The role of the audit committee (in overseeing financial reporting processes and mitigating opportunistic managerial behavior) derives from information asymmetries and resultant agency issues inherent in the separation of ownership and managerial control (Jensen and Meckling, 1976; Fama and Jensen, 1983). Prior studies of audit committee effectiveness tend to focus on non-financial firms. However, since the global financial crisis of 2007-2009, an evidence base for the financial services industry has emerged.

Prior literature examines audit committee effectiveness in terms of authority, size, independence, financial literacy of members and frequency of meetings (DeZoort et al., 2002; Bédard and Gendron 2010; Cohen et al., 2014). Research has sought to measure audit committee effectiveness in delivering on the core task of monitoring the financial reporting process with reference to a variety of proxies of reporting quality. Early research investigates differences in the financial reporting quality of firms with and without an audit committee. McMullen (1996) finds that the presence of an audit committee is associated with higher financial reporting quality. Following the emergence of high-profile accounting scandals during the early 2000s, research shifted toward investigating the characteristics of effective audit committees. Abbott et al., (2000) find a lower incidence of accounting misbehaviour (less SEC accounting enforcement sanctions) at firms with independent members on their respective audit committees. Abbott and Parker (2000) find that independent audit committee members demand a high level of audit quality in order to assure against potential reputational losses arising from poor financial reporting quality. Klein (2002) finds that audit committee independence constrains earnings manipulation.

A lack of financial literacy of audit committee members may impact on financial reporting quality. Members of audit committees may emanate from a wide variety of backgrounds and may not have the requisite experience for effective accounting oversight. Moreover, the audit committee's oversight role may be criticized if external auditors believe that the audit committee does not have the collective knowledge necessary to understand financial reporting matters (Cohen et al., 2002). Abbott et al., (2004) present evidence suggesting a lower incidence of financial re-statements for firms with audit committees with

at least one financial expert.⁹ Schmidt and Wilkins (2013) find that financial reporting quality (measured as the time elapsed between the discovery of a firm's need for a restatement and the actual disclosure of the restatement) is higher for firms with more financial experts. Badolato et al. (2014) find that firms with audit committees comprising members with financial expertise are less likely to engage in earnings management.

Bédard et al. (2004) investigate whether member independence and financial literacy impact financial reporting quality. Using a sample of firms with high and low levels of discretionary accruals, the authors find that audit committee member independence and financial literacy reduce the likelihood of aggressive earnings management. DeFond et al. (2005) find a positive market reaction to the appointment of audit committee members with higher levels of financial expertise. Karamanou and Vafeas (2005) find more accurate earnings forecasts for firms with audit committees comprising independent and financial literate members. Using accruals quality (estimated using cash flow realizations) as a proxy for financial reporting quality, Dhaliwal et al. (2010) find that financial reporting quality is higher in firms with a larger proportion of accounting experts and independent members on the audit committee.

Accounting related scandals in the United States and elsewhere and the increased numbers of financial re-statements highlight the importance of more conservative financial reporting and effective internal control (Krishnan, 2005; Ettredge et al., 2012; Chen et al. (2014). Using a sample of firms from Standard and Poor's 500, Krishnan and Visvanathan (2008) find that the appointment of individuals with prior financial expertise to audit committees leads to increased accounting conservatism (which ensures the negative news is reflected in earnings announcement more quickly than good news). Hoitash et al. (2009) find that there is a lower incidence of material weakness disclosures in internal control for firms with financial experts on their audit committees.

The effectiveness of audit committees in monitoring the financial reporting quality depends on resources deployed. Evidence regarding whether greater resources afforded to larger audit committees are more effective in ensuring financial reporting quality is rather mixed. Xie et al. (2003) and Abbott et al. (2004) find no evidence that the size of audit committees has an impact on accruals and restatements, while Bédard et al. (2004) find that

⁹ Evidence also suggests that industry expertise (Cohen et al., 2014), legal expertise (Krishnan et al., 2011) and task-specific expertise (Shepardson, 2019) of members of audit committees are positively related to measures of financial reporting quality.

the size of audit committees is associated with less aggressive earnings management behavior.

Prior research also investigates whether the frequency of audit committee meetings matters for financial reporting quality. With the exception of studies by Sharma and Iselin (2012) and Krishnan and Visvanathan (2008) who find that the number of audit committee meetings is associated with more restatements and more internal control weakness, prior research shows more frequent meetings are associated with: less aggressive earnings management (Bédard et al., 2004); higher external audit quality (Abbott and Parker, 2000; Abbott and Parker, 2002); fewer restatements (Abbott et al., 2004); higher quality of earnings (Vafeas, 2005); and less fraud (Farber, 2005). Bratten et al. (2022) find that financial reporting quality improves when audit committees are more proactive in monitoring external audits.

3.3 The Role of Audit Committees in Financial Firms

In the financial sector, there is a paucity of evidence regarding the effectiveness of audit committees. According to recent guidelines, bank audit committee responsibilities include: reviewing accounting estimates, financial reporting judgements, and financial statement disclosures; monitoring and disciplining management accountable for addressing identified deficiencies (such as violations of law or regulation); overseeing internal control systems and the internal and external audit functions; and meeting with bank examiners at least once each supervisory cycle (Office of the Comptroller Currency (OCC), 2016; Federal Reserve Board, 2017b). Given the complexity and size of banking organizations, Cornett et al. (2009) use the size and the frequency of meetings of audit committees as instrumental variables in order to investigate the prevalence of earnings management behaviour in large US bank holding companies in the period prior to the passing of the Sarbanes-Oxley Act of 2002. The authors find that effective audit committees constrain earnings manipulation. Conditions leading up to and surrounding the global financial crisis prompted increased concerns regarding excessive bank risk-taking. Using a sample of publicly listed financial institutions, Sun and Liu (2014) find that banks with audit committees comprising members with long tenure are associated with lower total and idiosyncratic risk. They also present evidence that banks with directors on their audit committees who also sit in other boards (inter-bank busyness) had higher total and idiosyncratic risk. In the case of the Dodd-Frank Act, Hines et al. (2015) document that the mandatory creation of a stand-alone risk committee is associated with higher audit fees. However, the authors present evidence

suggesting that when members of audit committees sit also on risk committees, there is a reduction in audit fees. Using a cross-country sample and including years before the financial crisis, Garcia-Sanchez et al. (2017) find that members of audit committees with prior financial expertise decrease bank risk-taking in countries where the bank regulatory system is weak, and investors feel unprotected. Beyond the oversight of financial reporting quality, members of audit committees supervise external audit quality. Recent evidence suggests that the incidence of earnings management at banks decreases when audit committees include former external auditors as members (Ittonen et al., 2018).

Overall, the extant literature finds that effective audit committees improve financial reporting quality. While there is an abundance of evidence for non-financial firms, research regarding the role of audit committees in ensuring financial reporting quality at banks remains somewhat limited. In the remainder of this paper, we investigate whether the separation of audit and risk committees (required by Section 165h of the Dodd-Frank Act) leads to an improvement in the financial reporting quality of US bank holding companies.

4. Research design

To estimate the importance of audit committee oversight on the financial reporting quality of banks, we focus on a specific policy change incorporated in Section 165h of the 2010 Dodd Frank Act. Section 165h requires publicly traded bank holding companies with assets exceeding \$10 billion to have separate committees for audit and risk oversight. Many of the provisions within the Dodd-Frank Act outline specific asset thresholds for compliance, often exempting smaller banks for which compliance is likely to be prohibitively costly. In the current setting, Section 165h focuses on publicly traded bank holding companies with assets exceeding \$10 billion. The Dodd-Frank Act was signed into law on 21st July 2010. The deadline for compliance with Section 165h for publicly traded bank holding companies whose asset size exceeding \$10 billion was 1st July 2015. The relevant asset size was that prevailing as of June 2014 (Federal Register, 2014).

Taking into consideration that banks have five years to comply with Section 165h, we follow a special procedure to form our sample of bank holding companies. Following prior literature, we begin with publicly traded bank holding companies with total assets exceeding \$10 billion after 2010 (Balasubramanyan et al., 2019). We identify 60 bank holding companies with assets exceeding \$10 billion as of June 2014. Of the 60 bank holding companies, eight have total assets less than \$10 billion in the pre-2014 period. While we

observe that these eight institutions exceed the \$10 billion threshold in the post 2010 period, our sample only includes the 52 bank holding companies with total assets exceeding \$10 billion in 2010.¹⁰

In order to identify affected (treated) and unaffected (control) institutions, we investigate whether bank holding companies have a joint audit and risk committee one year prior to the passage of Section 165h of the Dodd Frank Act. Specifically, bank holding companies with a joint audit and risk committee in 2009 are assigned to a treatment group, while counterparts that have both an independent risk committee and an independent audit committee are assigned to the control group. This results in seven more bank holding companies being excluded from our sample, because they have a joint risk committee with another committee. We identify 45 bank holding companies, of which 23 were non-compliant as of the signing of the law, and these institutions constitute the treatment group. The remaining 22 bank holding companies, which do not have a joint audit and risk committee, when the law was passed, constitute our control group.

In order to construct the discretionary loan loss provisions measure, we follow Beatty and Liao (2014) and estimate the following model:

$$LLP_{it} = \beta_0 + \beta_1 \Delta NPA_{it+1} + \beta_2 \Delta NPA_{it} + +\beta_3 \Delta NPA_{it-1} + \beta_4 SIZE_{it-1} + \beta_5 \Delta LOANS_{it} + \beta_6 ALW_{it-1} + \beta_7 CO_{it} + \beta_8 CSRET_{it} + \beta_9 \Delta GSP_{it} + \beta_{10} \Delta UNEMP_{it} + \varepsilon_{it}$$

$$(1)$$

Where LLP_{it} represents loan loss provisions, ΔNPA_{it} represents the change in nonperforming assets between year *t* and *t*-1, $SIZE_{it-1}$ is the natural logarithm of total assets in year *t*-1, $\Delta LOANS_{it}$ is the change in total loans over the year, ALW_{it-1} represents loan loss allowances in year *t*-1 and CO_{it} is net charge-offs in year *t*. The model also includes $CSRET_{it}$, ΔGSP_{it} and $\Delta UNEMP$. These variables denote the return on the Case-Shiller Real Estate Index, the change in gross state product and the change in the rate of state unemployment, respectively.¹¹ Full definitions of these aforementioned variables are provided in panel B of

¹⁰ In Section 7 we consider whether our findings are sensitive to the inclusion of those banks that grew in size to exceed the regulatory threshold of \$10 billion as of June 2014. By way of preview our findings remain qualitatively unchanged.

¹¹ We include Δ GSP to capture effects of macroeconomic conditions on loan loss provisions (Laeven and Majnoni, 2003; Bikker and Metzemakers, 2005; Fonseca and Gonzalez, 2008). Our results are consistent with pro-cyclical loan loss provisioning behavior.

Table 1, summary statistics are tabulated in Table A1 while the results obtained from estimating Equation (1) are presented in Table A2.

The absolute value of residuals from Equation (1) are discretionary loan loss provisions. We posit that a reduction in the amount of discretionary loan loss provisions for affected banks after the enactment of Section 165h of Dodd-Frank Act would imply higher financial reporting quality. This is consistent with our prediction that when audit committees focus on the core function of reporting oversight, financial reporting quality is improved via a reduction in discretionary loan loss provisions.

To investigate the impact of the separation of audit and risk committees on financial reporting quality, we use a difference-in-differences approach. This approach compares the difference in reporting quality between the treated and control bank holding companies in the pre- and post- event periods. We estimate a model of the form:

$$Y_{i,t} = \beta_1 (Affected_i * Post Event_t) + \delta X_{i,t} + v_i + \gamma_t + \varepsilon_{i,t},$$
(2)

where i indexes bank and t indexes time. $Y_{i,t}$ denotes the financial reporting quality measure. *Affected*_i is a dummy variable equal to one if a bank holding company has a joint audit and risk committee one year prior to the passage of Section 165h of the Dodd Frank Act., and zero otherwise. *Post Event*_t is a dummy variable for the post-treatment period. This variable takes the value of one for years 2010 onwards, and zero otherwise. *Affected*_i * *Post Event*_t is an interaction term which takes the value of one if the bank holding company is forced to separate audit and risk committees after Section 165h came into force, and zero otherwise. Our coefficient of interest is β_1 , which represents the impact of the separation of audit and risk committees on the financial reporting quality of bank holding companies.

 $X_{i,t}$ represents a vector of bank holding company-level control variables that vary across bank holding companies and over time. These control variables include size, dividends, the one-year lag of loan loss provisions, capital, earnings before taxes and loan loss provisions and the one-year ahead change in earnings before taxes and loan loss provisions. Prior evidence suggests that larger banks produce higher quality financial reporting information (Altamuro and Beatty 2010; Delis et al., 2018). Dividends could also influence financial reporting quality, given the likely negative association with higher earnings persistence (Lawson and Wang, 2016). Prior research also suggests that one-year lag loan loss provisions are associated with financial reporting quality (Kanagaretnam et al., 2010; Jiang et al., 2016).

We also control for possible relationships between discretionary loan loss provisions and capital management, earnings smoothing and signaling. A negative association between discretionary loan loss provisions and capital indicates that bank managers use loan loss provisions to manage capital (Ahmed et al., 1999; Kanagaretnam et al., 2004; Curcio and Hasan, 2015). A positive association between discretionary loan loss provisions and earnings before taxes and loan loss provisions is consistent with income smoothing behavior (Liu and Ryan, 2006; Fonseca and Gonzalez, 2008; Curcio and Hasan, 2015). Positive correlations between discretionary loan loss provisions and changes in future earnings are consistent with signaling behavior (Ahmed et al., 1999; Curcio and Hasan, 2015).

Finally, we introduce six additional covariates, which prior literature suggests are important drivers of audit committee effectiveness. These comprise the size of the audit committee, the financial expertise of members of the audit committee, the frequency of audit committee meetings, the intra-bank busyness, the inter-bank busyness and the tenure of audit committee directors. Table 1 provides a full list of variables included in the model along with their respective definitions. Equation (2) also includes time fixed effects, γ_t , as well as bank fixed effects v_i , which control for unobserved bank heterogeneity. Standard errors $\varepsilon_{i,t}$, are clustered at the bank level to account for possible autocorrelation.

[Insert Table 1 near here]

5. Data and summary statistics

We construct our dataset from three main data sources. At individual bank holding company level, we collect data from the S&P Global Market Intelligence (previously SNL financial) database. Information on analyst forecasts and on corresponding actual earnings per share is collected from Institutional Brokers Estimate System (I/B/E/S). Our sample period spans from 2007 to 2016. This period straddles the introduction of Section 165h of Dodd-Frank Act in 2010. In order to identify the structure and characteristics of audit committees, we assemble a hand collected dataset using the annual (10-K) reports and definitive proxy (DEF 14A) statements submitted by banks at the Securities and Exchanges Commission (SEC) and made available via the EDGAR database. Finally, macroeconomic variables are collected from the Bureau of Economic Analysis.

Assigning institutions to treatment and control groups as described in Section 4 results in a final sample of 45 unique bank holding companies (23 treated and 22 control banks) with 422 bank holding company-year observations due to missing values. We classify observations from 2007-2009 and 2010-2016 as the pre- and post-treatment periods respectively. A key identifying assumption underlying our estimation strategy is that in the absence of Section 165h the average discretionary loan loss provisions made by both affected and unaffected banks would have evolved in a parallel fashion. This implies a similar trend for the outcome variable for both the treatment and control group in the pre-treatment period. To mitigate any concerns regarding the violation of the parallel trend assumption, we construct a control group using an entropy balanced matching procedure, which exhibits a trend in discretionary loan loss provisions during the pre-treatment period similar to that of the treated group. Unlike propensity score matching, entropy balancing reweights observations in the control group such that the distribution of the control banks matches that of the affected banks (Hainmueller, 2012). Moreover, this is achieved without discarding observations in the process.

Table 2 provides descriptive statistics for the main variables for the banks in the treated group, and the banks in the control group before and after matching, spanning the entire period of investigation. Panels A and B tabulate these descriptive statistics for the period before and after the passage of Section 165h of the Dodd Frank Act, while Panel C tabulates the evolution of the outcome variable for the pre-treatment period. The descriptive statistics indicate that the trend in discretionary loan loss provisions is similar between affected and unaffected banks after the matching procedure in the pre-treatment period. Figure 1 provides a graphical illustration of this finding. It depicts the evolution of discretionary loan loss provisions for bank holding companies in the treatment period), discretionary loan loss provisions of affected and unaffected banks follow similar paths. However, from 2010 (the year that Section 165h of Dodd-Frank Act came into effect), we observe diverging trends for the affected and control bank holding companies. These findings lend support to the notion that the parallel trend assumption is not violated in our setting (the results of more formal tests are reported in Section 7).

[Insert Table 2 near here] [Insert Figure 1 near here]

6. Results

6.1 Baseline results

Table 3 presents the results of estimating Equation (2) using discretionary loan loss provisions as a dependent variable. Column 1 presents results of our baseline regression. We find a significantly negative coefficient on our primary variable of interest (*Affected x Post Event*) at the 1% level. The point estimate suggests that the amount of discretionary loan loss provisions of affected bank holding companies declines by 22 basis points. This decline is also economically significant, given that this represents almost 85% of the discretionary loan loss provisions made by the average bank holding company in the sample (which is equal to 26 basis points). Affected bank holding companies reduce loan loss provisions by \$257 million.¹² This suggests that relative to unaffected counterparts, bank holding companies affected by the introduction of Section 165h of Dodd-Frank Act are less likely to use discretionary loan loss provisions. This is in line with prior evidence, which suggests that lower discretionary loan loss provisions imply higher financial reporting quality (Wahlen 1994; Kanagaretnam et al., 2004; Fonseca and Gonzales 2008; Kanagaretnam et al., 2010; DeBoskey and Jiang, 2012).

In our analysis thus far, we have used the absolute value of residuals to measure discretionary loan loss provisions. We augment this analysis to investigate the direction of change in discretionary loan loss provisions. Prior evidence suggests that managers have incentives to overstate earnings using negative discretionary loan loss provisions (Huizinga and Laeven, 2012; Norden and Stoian, 2014; Jiang et al., 2016). On the other hand, banks may engage in over-provisioning in order to accumulate additional resources that could be used to boost their earnings at a later date (Norden and Stoian, 2014). Following prior literature (Danisewicz et al., 2021), we measure positive and negative discretionary loan loss provisions in Equation (1) and use these derived estimates as dependent variables in reestimated versions of Equation (2). The results are presented in columns 2 and 3 of Table 3, respectively. In the case of positive discretionary loan loss provisions, we find a negative and

¹² Economic significance is calculated by multiplying the coefficient of interaction term by the amount of total loans of the average bank in our sample.

significant coefficient on our variable of interest (*Affected* x *Post Event*) at the 1% level, while for negative discretionary loan loss provisions, the interaction term does not enter the regression with a statistically significant coefficient. These results suggest that the decrease in the absolute value of discretionary loan loss provisions is driven by the decline in incomedecreasing provisioning.

Turning to our control variables, we focus our discussion on the baseline specification (column 1) in Table 3. We control for the effectiveness of audit committees using: audit committee size; extent of membership financial expertise; frequency of meetings, percentage of busy audit committee directors (both intra- and inter-bank busyness); and average tenure of directors (Bédard et al., 2004; Yang and Krishnan, 2005; Krishnan and Visvanathan, 2008; Dhaliwal et al., 2010, Sun and Liu, 2014). Audit Committee Size enters the regression with positive, but insignificant coefficient, while Financial Expert, Intra-bank busyness and Interbank busyness enter the regression with negative, but insignificant coefficients. Finally, *Meetings* and *Tenure* enter the regression with positive and significant coefficients. Regarding our bank holding company level control variables, Size enters the regression with a negative, but statistically insignificant coefficient, while *lag_LLP* and *Dividends* enter the regression with a positive, but insignificant coefficient. We also find that better capitalized bank holding companies make lower discretionary loan loss provisions (as indicated by the negative and statistically significant coefficient on capital). This is in line with prior literature, which suggests that banks use discretionary loan loss provisions to satisfy regulatory capital requirements (Ahmed et al., 1999; Curcio and Hasan, 2015). Furthermore, EBTLLP enters the regression with a negative, but statistically insignificant coefficient. Finally, we do not find evidence of the signaling hypothesis as indicated by the negative and statistically coefficient in future earnings.

[Insert Table 3 near here]

Overall, the results of our empirical analysis suggest that affected bank holding companies improve financial reporting quality following the introduction of Section 165h of Dodd-Frank Act.

6.2 Underlying mechanisms

The results of our empirical analysis (described in Section 6.1) suggest that affected bank holding companies improve financial reporting quality following the introduction of Section 165h of Dodd-Frank Act. The separation of the joint audit and risk committee into two independent committees after the introduction of Section 165h of the Dodd-Frank Act decreases the scope of responsibilities of the audit committee members, thus allowing for increased focus on the core function of financial reporting quality oversight. This suggests that when members of the audit committees are not overloaded, they are more effective at their core functions with subsequent improvement in financial reporting quality. In this section we explore the mechanisms through via which this outcome is realised. We identify five such channels based on audit committee characteristics including size, financial expertise, frequency of meetings, intra- and inter-bank busyness and tenure that relate to the audit committee's effectiveness.

The size of the audit committee is an important factor in enhancing financial reporting quality. This is due to the fact that larger audit committees can rely on a wider knowledge base and varied expertise of members. As such they can undertake their role more effectively (Vafeas, 2005). If audit committee size affects financial reporting quality, then any resultant impact of the separation of audit and risk committees following the introduction of Section 165h of Dodd-Frank Act should be higher among affected bank holding companies with larger audit committees.

Regulators have also shown a considerable interest in the financial expertise of audit committee members. In the United States, following the Sarbanes-Oxley Act (2002) audit committees ought to comprise at least one financial expert, while the other members need to be financially literate. Prior research suggests that the presence of financial expertise on the audit committee is positively associated with financial reporting quality (Xie et al., 2003; Abbott et al., 2004; Bedard et al., 2004). Therefore, we would expect the impact of the introduction of Section 165h of Dodd-Frank Act on financial reporting quality to be higher among bank holding companies with a higher proportion of audit committee members with financial expertise.

The activity level of audit committees plays an important role in overseeing and monitoring the financial reporting process. It is argued that frequent audit committee meetings play a crucial role in addressing agency problems between management and various stakeholders. A number of prior studies provide evidence in support of the importance currently attached to the level of activity undertaken by the audit committee (Abbott et al., 2003; Vafeas, 2005; Hoitash et al., 2009; Sharma et al., 2009; Engel et al., 2010). These studies suggest that more frequent audit committee meetings result in less aggressive earnings management. Therefore, if the separation of audit and risk committees affects financial reporting quality via a change in the frequency of audit committee meetings, then any resultant impact of Section 165h of Dodd-Frank Act should be higher among bank holding companies with a higher frequency of audit committee meetings.

The relationship between multiple directorships and firm performance remains a major concern for regulators and academics. Existing evidence shows that serving on numerous boards —inter-bank busyness (Shivdasani and Yermack, 1999; Fich and Shivdasani, 2006) or on numerous committees on the same board —intra-bank busyness (Chen and Wu, 2016) can result in busy and overstretched directors that may not be effective monitors of corporate management. Dhaliwal et al. (2010) document that audit committee members with fewer directorships are associated with less earnings management. In light of these arguments, we expect that if the separation of audit and risk committees affects financial reporting quality via a change in the composition of audit committee membership, then any resultant impact of Section 165h of Dodd-Frank Act on financial reporting quality should be higher for bank holding companies with audit committees comprising less busy directors.

The length of tenure of audit committee directors is likely to determine the effectiveness via which audit committees perform their monitoring role. Prior literature suggests that companies with audit committees where members have a long tenure exhibit lower financial reporting quality (Vafeas, 2005; Dhaliwal et al., 2010). Therefore, we expect that the introduction of Section 165h will have a higher impact on financial reporting quality among banks with higher tenure audit committee members.

To determine which of these aforementioned non-mutually exclusive mechanisms explain how the separation of audit and risk committees results in an improvement in financial reporting quality oversight, we re-estimate Equation (2) after sequentially replacing the respective audit committee characteristics as dependent variables. We express the dependent variables (Size, Financial Expert, Meetings and Tenure) in logarithmic form in order to make the estimated coefficients comparable across the different models. The results are presented in columns 1-6 of Table 4. The only cases that the interaction term enters the regression with a significant coefficient are in columns 4 and 5, where *Intra-bank busyness* and *Inter-bank busyness* are the dependent variables, respectively. The sign of the coefficient is negative in both cases. Moreover, the effect is more prominent in the case of *Intra-bank busyness* with the magnitude of the reported coefficient on the interaction term is three times higher than that of the *Inter-bank busyness*. These findings suggest that reducing audit committee members' busyness in general and *Intra-bank busyness* in particular can be an effective channel for improving financial reporting quality.

[Insert Table 4 near here]

6.3 Alternative explanation

Our results, thus far suggest that Section 165h reduced discretionary loan loss provisions at affected banks due to the increased focus of the audit committee on the core function of financial reporting quality oversight. This was achieved via the establishment of independent risk committees resulting in the separation of joint audit and risk committees. However, one could argue that this documented reduction in discretionary loan loss provisions of affected banks was the result of audit committee members being prudent, while in charge of bank risk oversight.

We test whether the level of risk at affected banks changes as a result of the introduction of Section 165h. We re-estimate Equation (2) using credit risk (measured by the ratio of non-performing loans to total loans) as the dependent variable.¹³ We expect credit risk to be higher in cases where risk oversight is undertaken by board members with a lack of requisite knowledge and expertise. In this case, we would expect audit committee members to over-provision. The results of this test, which are tabulated in column 7 of Table 4, show that the level of credit risk at affected banks remains unchanged between the pre-and post-implementation period relative to the banks in the control group. This refutes the alternative explanation of our findings.

7. Additional tests

In this section we investigate the robustness of our main findings by conducting several tests that support the causal interpretation of the results obtained from our analysis

¹³ Our results remain unaltered when using net charge offs as an alternative proxy for credit risk.

above. Furthermore, we explore possible confounding effects that could drive our estimated results.

Falsification tests

The analysis presented thus far has shown that bank holding companies affected by the separation of audit and risk committees through the implementation of Section 165h of the Dodd-Frank Act improve their financial reporting quality. However, the validity of the difference-in-differences estimation requires that in the absence of the treatment, financial reporting quality for both treated and control bank holding companies follow the same behavior. This is referred to as the parallel trend assumption (Abadie, 2005). We complement the initial investigation of the parallel trend assumption reported in Section 5 (Table 2 Panel C) by conducting two additional tests. First, we conduct a placebo test, which falsely varies the timing of the introduction of Section 165h. Following Ignatowski and Korte (2014), we extend our dataset to cover the period 2002 to 2006, which we define as the pre-placebo period. In order to investigate the effect of a placebo treatment, we assume falsely that the separation of the audit and risk committees took place in 2008 rather than in 2010. Therefore, the sample used for this test covers the period 2002 through 2009. The results of this test are displayed in column 1 of Table 5. The coefficient on the interaction term (Affected x *Placebo Post Event*) is statistically insignificant. Second, we investigate the dynamics of financial reporting quality around the introduction of Section 165h, by plotting the annual average gap in the discretionary loan loss provisions of the treated banks relative to their unaffected counterparts. Figure 2 shows that the trend in the discretionary loan loss provisions of the treated banks is statistically not different from that of the control group during the pre-treatment period. This is another indication that the parallel trend assumption is not violated in our sample. Both Figure 2 and the results of the placebo test suggest that the parallel trend assumption is not violated for our analysis and further supports the causal interpretation of our main findings.

Moreover, our analysis includes several time-varying control variables to mitigate the risk of the parallel trend assumption violation. Such inclusion, however, introduces the risk of biasing the estimated treatment effect (Atanasov and Black, 2016). In order to ensure that our results do not suffer from such a bias, we re-estimate Equation (2) without including time-varying control variables. The results of this analysis, which are reported in column 2 of Table 5, show that the magnitude of the coefficient of interest (*Affected x Post Event*) remains virtually unchanged and thus our main conclusions continue to hold. In addition, we utilize a

coefficient stability test in order to investigate possible omitted variable bias in our estimates. The test (proposed by Oster, 2019) quantifies how much stronger the effect of unaccounted (relative to accounted) factors influencing financial reporting quality has to be in order to obtain a zero difference-in-differences estimate. The results of this test indicate that the effect of unobservables would have to be twice as important as the included covariates for the introduction of Section 165h of the Dodd-Frank Act to have no effect on financial reporting quality of banks.¹⁴ This implies that our findings are not driven by unobservable characteristics.

[Insert Figure 2 near here] [Insert Table 5 near here]

Serial correlation

A concern regarding the validity of a difference-in-differences estimation arises if the standard errors are serially correlated. In such cases, reported standard errors could be biased downwards. To investigate concerns regarding serial correlation, we follow Bertrand et al. (2004), and collapse our sample into two time periods. Specifically, we average our variables before and after the separation of audit and risk committees by the introduction of Section 165h of the Dodd-Frank Act. The results of this exercise, which are reported in column 3 of Table 5 confirm that our findings are not driven by serial correlation as the coefficient of interest (*Affected x Post Event*) retains its significance. *Sensitivity analysis*

Another potential driver of our results could arise from the requirement that bank holding companies had to establish independent risk committees after the introduction of Section 165h of the Dodd-Frank Act. Delis et al. (2018) find that enforcement actions related to risk issues improves the accounting quality of US commercial banks. If this is the case and treated bank holding companies display an improved financial reporting quality because of the presence of an independent risk committee and hence a better risk management policy, this could confound our baseline results. In order to investigate this possibility, we re-

¹⁴ Using Oster (2019)'s terminology, we find that $\delta = 2.01$. This result assumes that the R_{max}^2 of the hypothetical regression which contains all unobservable factors of financial reporting quality is the product of the R^2 of the regression including the observable factors multiplied by 1.3. Specifically, we use $R^2 = 0.274$ reported in column 1 of Table 3 in order to obtain $R_{max}^2 = 1.3 * 0.274$.

estimate Equation (2) using as a treated group, bank holding companies that have a joint risk with another committee (compliance committee, capital committee, examination committee, etc.), and as a control group bank holding companies that already had an independent risk committee prior to the introduction of Section 165h of the Dodd-Frank Act.¹⁵ Specifically, in Equation (2) we replace *Affected* with *Joint Risk Committee*, a dummy variable that captures the distinction between the treated and control group described above. Column 1 of Table 6 reports the results of this analysis. The coefficient of the interaction term (*Joint Risk Committee* x *Post Event*) is statistically insignificant. This finding implies that only bank holding companies that have a joint audit and risk committee before the introduction of Section 165h of the Dodd-Frank Act show an improvement in financial reporting quality. This further confirms that the observed improvement in the financial reporting quality of the affected bank holding companies is driven by the increased focus of the audit committee.

We also investigate whether our results are robust to the selection of our sample of bank holding companies. Our findings are based on those bank holding companies that meet the size threshold set by Section 165h in 2010 when the Dodd-Frank Act was signed into law. After 2010, and during the ensuing five-year period these banks were given to comply with the Act, eight more banks grew to exceed the \$10 billion asset size threshold, and thus had to comply with the Act. In order to ensure that our findings are not driven by the exclusion of these banks we re-estimate Equation (2) with an expanded sample that includes the eight banks with total assets exceeding \$10 billion in the post 2010 period.¹⁶ The results of this analysis, which are reported in column 2 of Table 6, suggest that our main findings remain qualitatively unaltered to the inclusion of these observations.

Finally, another critical issue is the exact timing of when affected bank holding companies split their audit and risk committees given that they had five years to comply with Section 165(h) of Dodd-Frank Act. Our main analysis is conducted under the assumption that the separation of audit and risk committees occurred in 2010 for affected bank holding companies. To alleviate concerns that our results are driven by this assumption, we check our data to identify the year that affected bank holding companies split their audit and risk committees following the introduction of Section 165(h) of Dodd-Frank Act. Scrutiny of the definitive proxy statements (DEF 14A) of affected bank holding companies reveals that five

¹⁵ Following the same data collection procedure as in our baseline analysis (Section 5), we identify seven bank holding companies that have a joint risk with another committee before the introduction of section 165h of the Dodd Frank Act.

¹⁶ Of these eight banks, five are classified as treated. The rest are unaffected by Section 165h and thus are classified as control banks.

bank holding companies split their audit and risk committees in 2010, four in 2011, one in 2012, nine in 2013, three in 2014 and one in 2015. Therefore, we re-estimate Equation (2) by replacing *Post Event* to take the value of one the year that a bank holding company splits the audit and risk committee onwards, and zero otherwise. The results of this test, which are reported in column 3 of Table 6, suggest that our main findings hold.

[Insert Table 6 near here]

Confounding factors

The validity of our approach would be undermined if factors other than the separation of audit and risk committee (following the introduction of Section 165h of the Dodd-Frank Act) are driving our main results. In 2009, the US government introduced the Troubled Asset Relief Program (TARP) to assist distressed banks and stabilise the banking industry. As a result, TARP recipient banks would be subject to more and stricter government supervision(Bouvard et al., 2015). Specifically, banks that received TARP funding are required to be more transparent about how the funds are being used. Therefore, the TARP program could have an impact on bank financial reporting quality similar to that observed following the separation of the audit and risk committee after the introduction of Section 165h of the Dodd-Frank Act. In order to investigate the robustness of our findings to this issue, we introduce the variable TARP and its interaction with Affected in Equation (2). TARP is defined as a dummy variable that is equal to one for the period each bank holding company in our sample receives funding from the US government. The results are presented in column 1 of Table 7. The coefficient of the interaction term (*Affected* x *TARP*) enters the regression negatively and statistically insignificant, while the coefficient on the interaction term of interest (Affected x Post Event) remains negative and statistically significant. This suggests that our main findings are not driven by the provisions of TARP.

[Insert Table 7 near here]

Another potential effect that could confound the results of our analysis are changes in the state corporate income tax rates facing banks in our sample. Tax rates changes have been shown to influence earnings management behaviour and hence financial reporting quality. For example, Dong and Xu (2019) find that accounting earnings are more likely to be managed upward via the use of discretionary accruals in response to state tax cuts. If statutory tax rates were reduced in states where bank holding companies in our control group are located around the same time as the introduction of Section 165h of the Dodd-Frank Act, then this could confound the impact of the separation of audit and risk committee on financial reporting quality. In order to check the robustness of our findings to changes in state taxes, we hand collect information on state level statutory corporate income tax rates (from the Tax Foundation) and re-estimate Equation (2) via the introduction of the variable *Tax* and its interaction with *Affected*. We define *Tax* as a dummy variable equal to one if there is a cut in state corporate income tax rate, and zero otherwise. The results are presented in column 2 of Table 7. The interaction term *Affected* x *Tax* enters the regression with a positive and statistically significant coefficient, while importantly the coefficient of the interaction term of interest (*Affected* x *Post Event*) remains negative and statistically significant. This suggests that our main findings are not confounded by coincident changes in state corporate income taxes.¹⁷

We also investigate stress tests as a potential confounder to our findings. Prior evidence suggests that banks manage financial performance in order to meet capital ratio targets (Cochrane 2014, Cornett et al., 2020 Fernandes et al., 2020). Therefore, we explore whether our findings are driven by the stress tests exercises rather than the introduction of Section 165h of the Dodd-Frank Act. In order to check the robustness of our findings to stress tests, we re-estimate Equation (2), by incorporating the variable *Stress* and its interaction with *Affected*. Following prior literature, we define *Stress* as a dummy variable that equals one for the years that bank holding companies are part of the stress test exercises (Fernandes et al., 2020). The results, which are tabulated in column 3 of Table 7, indicate that the interaction term *Affected x Stress* is statistically insignificant, while the interaction term of interest *Affected x Post Event* retains its sign and significance. These results suggest that stress tests do not drive our main findings.

Alternative measures of financial reporting quality

We also investigate the sensitivity of our findings to alternative measures of financial reporting quality. Following prior literature, we compute small positive earnings changes and surprise avoidance. Small positive earnings changes capture the tendency of banks to manage earnings (so as to avoid reporting a negative change in earnings). We isolate

¹⁷ The results are robust to the use of an alternative proxy for capturing tax burden variation across states, namely the tax component of the Economic Freedom of North America index. The Economic Freedom index has been shown to correlate with US bank performance (Chortareas et al., 2016).

all bank holding company-years with a marginally positive earnings change. Prior evidence for non-financials suggests that firms that report small marginal positive income increases seek to avoid conveying bad news to investors (Burgstahler and Eames, 2003). Following Vafeas (2005), *Small Positive Earnings Changes* is defined as an indicator variable that equals one if there is a positive change of up to 1.5 percent over last year's net income divided by total assets, and zero otherwise. The surprise avoidance measure is also motivated based on prior evidence, which suggests that managers manipulate earnings upwards in order to avoid negative earnings surprises and meet analyst expectations (Matsumoto, 2002). Again, following Vafeas (2005), *Surprise Avoidance* is an indicator variable that equals one if an earnings surprise is between 0.00 and 0.04 cents over the consensus (median) analyst forecast, calculated as the last forecast prior to the announcement of annual earnings, and zero otherwise.

Table 8 shows that the interaction term *Affected x Post Event* is significantly and negatively associated with these alternative measures of financial reporting quality. This further illustrates the robustness of our main findings.

[Insert Table 8 near here]

8. Conclusion

In the aftermath of corporate governance and risk management failures during the global financial crisis of 2007-2009, US policymakers expressed concerns that audit committees did not have sufficient capacity to execute core responsibilities, leading to a deterioration in financial reporting quality. Resultant regulatory reforms embodied in Section 165h of the Dodd-Frank Wall Street Reform and Consumer Protection Act sought to tackle such issues by stipulating that large publicly traded bank holding companies should have separate, rather than joint audit and risk committees.

Against this backdrop, we investigate the importance of the audit committee for the financial reporting quality of US bank holding companies. The fact that some bank holding companies already had separate audit and risk committees prior to the enactment of the Section 165h of the Dodd-Frank Act allows us to identify a treated and control group of institutions and conduct a difference-in-differences analysis to assess the impact of policy reforms on the financial reporting quality (measured by discretionary loan loss provisions) of affected bank holding companies relative to unaffected counterparts.

The results of the empirical analysis presented in this paper suggest that financial reporting quality significantly improves following the introduction of Section 165h of the Dodd-Frank Act. Specifically, there is a significant reduction in earnings management at bank holding companies via a reduction in discretionary loan loss provisions. Consistent with the notion that directors who serve on many committees on the same board experience a reduced ability to be effective monitors of corporate management, we further find that this reduction in discretionary loan loss provisions is stronger for banks with fewer busy directors. An exhaustive series of additional tests confirm our findings.

Consequently, our results are of relevance for policymakers tasked with monitoring the impact of accounting based regulatory reforms on the behavior and performance of large banks. Specifically, the evidence presented in this paper suggests that active policy reforms aimed at increasing the focus of audit committees lead to improvements in financial reporting quality and resultant information available to industry stakeholders.

References

- Abadie, A., 2005. Semiparametric difference-in-differences estimators. *Review of Economic Studies*, 72(1), 1–19.
- Abbott, L. J., Park, Y., Parker, S., 2000. The effects of audit committee activity and independence on corporate fraud. *Managerial Finance*, 26(11), 55–67.
- Abbott, L. J., Parker, S., 2000. Auditor selection and audit committee characteristics. *Auditing: A Journal of Practice and Theory*, 19(2), 46–66.
- Abbott, L. J., Parker, S., 2002. Audit committee characteristics and auditor switches. *Research in Accounting Regulation*, 15, 151-166.
- Abbott, L.J., Parker, S., Peters, G.F., Raghunandan. K., 2003. The association between audit committee characteristics and audit fees. *Auditing: A Journal of Practice and Theory*, 22(2), 17–32.
- Abbott, L.J., Parker, S., Peters, G.F., Raghunandan, K. 2003. An empirical investigation of audit fees, non-audit fees and audit committees. *Contemporary Accounting Research*, 20(2), 215–234.
- Abbott, L.J., Parker, S., Peters, G.F., 2004. Audit committee characteristics and restatements. *Auditing: A Journal of Practice and Theory*, 23(1), 69-87.
- Aebi, V., Sabato, G., Schmid, M., 2012. Risk management, corporate governance, and bank performance in the financial crisis. *Journal of Banking & Finance*, 36(12), 3213–3226.
- Agenor, P.R., Zilberman, R. 2015. Loan loss provisioning rules, procyclicality, and financial volatility. *Journal of Banking & Finance*, 61, 301-315.
- Ahmed, A.S., Takeda, C., Thomas, S., 1999. Bank loan loss provisions: A re-examination of capital management, earnings management and signaling effects. *Journal of Accounting and Economics*, 28 (1), 1-25.
- Akhigbe, A., Martin, A.D., Whyte, A.M., 2016. Dodd-Frank and risk in the financial services industry. *Review of Quantitative Finance and Accounting*, 47, 395-415.
- Allen, F., Carletti, E., Gu, X., 2019. The roles of banks in financial systems, in A. Berger, P. Molyneux and J.O.S. Wilson (eds.) Oxford Handbook of Banking, 3rd Edition. Oxford: Oxford University Press.
- Altamuro, J., Beatty, A., 2010. How does internal control regulation affect financial reporting quality? *Journal of Accounting and Economics*, 49(1-2), 58-74.
- American Institute of Certified Public Accountants (AlCPA). 1999. SAS No. 89. Audit Adjustments. New York: AlCPA.

- Ashraf, M., Choudhary, P., Jaggi, J., 2020. The scope of audit committee oversight and financial reporting reliability: are audit committees overloaded? Available at SSRN: https://ssrn.com/abstract=3433389
- Atanasov, V.A, Black, B.S., 2016. Shock-based causal inference in corporate finance and accounting research. *Critical Finance Review*, 5, 207-304.
- Avraham, D., Selvaggi, P., Vickery, J., 2012. A structural view of bank holding companies. *Federal Reserve Bank of New York Economic Policy Review* 18(2), 65-82.
- Badolato, P.G., Donelson, D.C., Ege, M., 2014. Audit committee financial expertise and earnings management: The role of status. *Journal of Accounting and Economics*, 58, 208-230.
- Barth, E.M., Gomez-Biscarri, J., Kasznik, R., Lopez-Espinosa, G., 2017. Bank earnings and regulatory capital management using available for sale securities. *Review of Accounting Studies*, 22, 1761-1792.
- Balasubramanyan, L., Daniel, N.D., Haubrich, J.G., Naveen, L., 2019. Causal impact of risk oversight functions on bank risk: Evidence from a natural experiment. *Federal Reserve Bank of Cleveland Working Paper* Number 19-01.
- Balasubramnian, B., Cyree, K.B., 2014. Has market discipline on banks improved after Dodd-Frank Act? *Journal of Banking & Finance*, 41, 155-166.
- Bates, E., Leclerc, R., 2009. Boards of directors and risk committees. *The Corporate Governance Advisor*, 17(6), 15-17.
- Beasley, M.S., Carcello, J.V., Hermanson, D.R., Neal, T.L., 2009. The audit committee oversight process. *Contemporary Accounting Research*, 26(1), 65-122.
- Beaver, W.H., Engel, E.E., 1996. Discretionary behavior with respect to allowances for loan losses and the behavior of security prices. *Journal of Accounting and Economics*, 22(1), 177-206.
- Beatty, A., Chamberlain, S., Magliolo, J., 1995. Managing financial reports of commercial banks: The influence of taxes, regulatory capital and earnings. *Journal of Accounting Research*, 33(2), 231-262.
- Beatty, A., Liao, S., 2011. Do delays in expected loss recognitions affect banks' willingness to lend? *Journal of Accounting and Economics*, 52(1), 1-20.
- Beatty, A., Liao, S., 2014. Financial accounting in the banking industry: A review of the empirical literature. *Journal of Accounting and Economics*, 58(2-3), 339–383.
- Bédard, J., Chtourou, S.M., Courteau, L., 2004. The effect of audit committee expertise, independence, and activity on aggressive earnings management. *Auditing: A Journal of Practice & Theory*, 23(2), 13-35.
- Bédard, J., Gendron, Y., 2010. Strengthening the financial reporting system: Can audit committees deliver? *International Journal of Auditing*, 14(2), 174-210.

- Berger, A., Molyneux, P., Wilson, J.O.S., 2020. Banks and the real economy: An assessment of the research. *Journal of Corporate Finance*, 62, 101513.
- Bertrand, M., Duflo, E., Mullainathan, S., 2004. How much should we trust difference-indifferences estimates? *Quarterly Journal of Economics*, 119 (1), 249-275.
- Bikker, J.A., Metzemakers, P.A., 2005. Bank provisioning behavior and procyclicality. *Journal* of International Financial Markets, Institutions and Money, 15(2), 141-157.
- Blue Ribbon Commission on improving the effectiveness of corporate audit committees. 1999. Report and recommendations of the Blue-Ribbon Committee on improving the effectiveness of corporate audit committees. *Business Lawyer*, 54(3), 1067–1095.
- Bouvard, M., Chaigneau, P., De Motta, A., 2015. Transparency in the financial system: Rollover risk and crises. *Journal of Finance*, 70(4), 1805-1837.
- Bouvatier, V., Lepetit, L., 2008. Banks' procyclical behavior: Does provisioning matter? *Journal of International Financial Markets, Institutions and Money*, 18(5), 513-526.
- Bouvatier, V., Lepetit, L., Strobel, F., 2014. Bank income smoothing ownership concentration and the regulatory environment. *Journal of Banking & Finance*, 41, 253-270.
- Bratten, B., Causholli, S., Sulcaj, V., 2020. Overseeing the external audit function: Evidence from audit committees' reported activities. *AUDITING: A Journal of Practice & Theory* (forthcoming).
- Burgstahler, D., Eames, M., 2003. Earnings management to avoid losses and earnings decreases: Are analysts fooled? *Contemporary Accounting Research*, 20(2), 253-294.
- Bushman, R.M., Smith, A.J., 2001. Financial accounting information and corporate governance. *Journal of Accounting and Economics*, 32(1-3), 237-333.
- Bushman, R.M., Williams, C., 2012. Accounting discretion, loan loss provisioning, and discipline of bank's risk taking. *Journal of Accounting and Economics*, 54(1), 1-18.
- Bushman, R.M., 2014. Thoughts on financial accounting in the banking industry. *Journal of Accounting and Economics*, 58(2-3), 384-395.
- Carcello, J., Neal, T.L., 2000. Audit committee composition and auditor reporting. *The Accounting Review*, 75(4), 453-467.
- Cetorelli, N., Goldberg. L., 2014. Measures of global bank complexity. *Federal Reserve Bank of New York Economic Policy Review*, 20(2), 107-126.
- Cetorelli, N., McAndrews, J., Traina, J., 2014. Evolution in bank complexity. *Federal Reserve Bank of New York Economic Policy Review*, 20(2), 85-106.
- Chen, K.Y., Elder, R.J., Hung, S., 2014. Do post-restatement firms care about financial credibility? Evidence from the pre- and post-SOX eras. *Journal of Accounting and Public Policy*, 33(2), 107-126.

- Chen, K.D., Wu, A., 2016. The structure of board committees. *Academy of Management Proceedings,*
- Chortareas, G., Kapetanios, G., Ventouri A., 2016. Credit market freedom and cost efficiency in US state banking. *Journal of Empirical Finance* 37, 173-185.
- Claypool, G., Tacket, J., Wolf, F., 2004. Sarbanes-Oxley and audit failure. A critical examination. *Managerial Auditing Journal*, 19(3), 340-350.
- Cochrane, J.H., 2014. Toward a run-free financial system, in Baily M.N., and J.B. Taylor (eds.) *Across the Great Divide: New Perspectives on the Financial Crisis*. Stanford, California: Hoover Institution Press.
- Coffee, J.C. (2005). A theory of corporate scandals: Why the USA and Europe differ. *Oxford Review of Economic Policy*, 21(2), 198–211.
- Cohen, J., Krishnamoorthy, G., Wright, A.M., 2002. Corporate governance and the audit process. *Contemporary Accounting Research* 19(4), 573-594.
- Cohen, J., Krishnamoorthy, G., Wright, A.M., 2010. Corporate governance in the post Sarbanes-Oxley era: Auditors' Experiences. *Contemporary Accounting Research*, 27 (3), 751-786.
- Cohen, J.R., Hoitash, U., Krisnamoorthy, G., Wright, A.M., 2014. The Effect of audit committee industry expertise on monitoring the financial reporting process. *The Accounting Review*, 89(1), 243-273.
- Cohen, L.J., Cornett, M.M., Marcus, A.J., Tehranian, H., 2014. Bank earnings management and tail risk during the financial crisis. *Journal of Money Credit Banking*, 46(1), 171-197.
- Copeland A. 2012. Evolution and heterogeneity among larger bank holding companies: 1994 to 2010. *Federal Reserve Bank of New York Economic Policy Review* 18(2), 83-93.
- Collins, J.H., Shackelford, D.A., Wahlen, J.M., 1995. Bank differences in the coordination of regulatory capital, earnings and taxes. *Journal of Accounting Research*, 33(2), 263-291.
- Cornett, M.M., McNutt, J.J., Tehranian, H., 2009. Corporate governance and earnings management at large US bank holding companies. *Journal of Corporate Finance*, 15(4), 412-430.
- Cornett, M.M., Minnick, K., Schorno, P.J., Tehranian, H., 2020. An examination of bank behavior around Federal Reserve stress tests. *Journal of Financial Intermediation*, 41, 100789.
- Crockett, A., 2002. Market discipline and financial stability. *Journal of Banking & Finance*, 26(5), 977-987.
- Curcio, D., Hasan, I., 2015. Earnings and capital management and signaling: The use of loan loss provisions by European banks. *European Journal of Finance*, 21(1), 26-50.

- Curcio, D., De Simone, A., Gallo A., 2017. Financial crisis and international supervision: New evidence on the discretionary use of loan loss provisions at Euro Area commercial banks. *British Accounting Review*, 49 (2), 181-193.
- Cyree, K.B., 2016. The effects of regulatory compliance for small banks around crisis-based regulation. *Journal of Financial Research*, 39(3), 215-245.
- Danisewicz, P., McGowan, D., Onali, E., Schaeck, K., 2021. Debtholder monitoring incentives and bank earnings opacity. *Journal of Financial and Quantitative Analysis*, 56(4), 1-38.
- DeBoskey, D.G., Jiang, W., 2012. Earnings management and auditor specialization in the postsox era: an examination of the banking industry. *Journal of Banking & Finance*, 36(2), 613-623.
- Dechow, P.M., Weili, G., Schrand, C., 2010. Understanding earnings quality: A review of the proxies, their determinants and their consequences. *Journal of Accounting and Economics*, 50(2-3), 344-401.
- DeFond, M.L., Hann, R.N., Hu, X., 2005. Does the market value financial expertise on audit committees of boards of directors? *Journal of Accounting Research*, 43(2), 153-193.
- Delis, M.D., Hasan, I., Iosifidi, M., Li, L., 2018. Accounting quality in banking: The role of regulatory interventions. *Journal of Banking & Finance*, 97, 297–317.
- DeZoort, F.T., Hermanson, D.R., Archambeault, D.S., 2002. Audit committee effectiveness: A synthesis of the empirical audit committee literature. *Journal of Accounting Literature*, 21 (1), 38–75.
- Dhaliwal, D., Naiker, V., Navissi, F., 2010. The association between accruals quality and the characteristics of accounting experts and mix of expertise on audit committees. *Contemporary Accounting Research*, 27(3), 787-827.
- Dong, G.N., Xu, L., 2019. Taxation and earnings management: Evidence from staggered changes in state corporate tax rates. Available at SSRN: https://ssrn.com/abstract=3367855
- El Sood, H.A., 2012. Loan loss provisions and income smoothing in US banks pre and post the financial crisis. *International Review of Financial Analysis*, 25, 64-72.
- Engel, E., Hayes, R., Wang, X., 2010. Audit committee compensation and the demand for monitoring of the financial reporting process. *Journal of Accounting and Economics*, 49(1-2), 136–154.

Ernst and Young. 2014. Leading Practices for Audit Committees. London: Ernst & Young.

- Ettredge, M., Huang, Y., Zhang, W., 2012. Earning restatements and differential timeliness of accounting conservatism. *Journal of Accounting Research*, 53(3), 489-503.
- Evans, L., Schwartz, J., 2004. The effect of concentration and regulation on audit fees: An application of panel data techniques. *Journal of Empirical Finance* 27, 130-144.

- Fama, E.F., Jensen, M.C., 1983. Separation of ownership and control. *Journal of Law & Economics*, 26(2), 301-325.
- Farber, D.B., 2005. Restoring trust after fraud: Does corporate governance matter? *Accounting Review*, 80(2), 539-561.
- Federal Register. 2014. Enhanced Prudential Standards for Bank Holding Companies and Foreign Banking Organizations; Final Rule.
- Federal Reserve Board. 2017(b). *Commercial Bank Examination Manual*. Washington DC: Federal Reserve Bank.
- Fernandes, M., Igan D., Pinheiro, M., 2020. March madness in Wall Street: (What) does the market learn from the stress tests? *Journal of Banking & Finance*, 112, 105250.
- Fich, E.M., Shivdasani, A., 2006. Are busy boards effective monitors, *Journal of Finance*, 61(2), 689–724.
- Financial Crisis Inquiry Commission. 2011. *Final Report of the National Commission on the Causes of the Financial and Economic Crisis in the United States*. Washington: Government Publishing Office
- Financial Stability Oversight Council. 2011. Report to the Congress on Prompt Corrective Action. Washington DC: FSOC.
- Financial Stability Oversight Council. 2012. Report to Congress on Study of a Contingent Capital Requirement for Certain Nonbank Financial Companies and Bank Holding Companies. Washington DC: FSOC.
- Financial Stability Oversight Council.2016. Study of the Effects of Size and Complexity of Financial Institutions on Capital Efficiency and Economic Growth. Washington DC: FSOC.
- Flannery, M.J., 2009. Market discipline in banking supervision, in Berger, A.N., Molyneux, P. and J.O.S. Wilson (eds.) *Oxford Handbook of Banking*. Oxford: Oxford University Press.
- Flannery, M.J., Kwan, S.H., Nimalendran, M., 2013. The 2007–2009 financial crisis and bank opaqueness, *Journal of Financial Intermediation*, 22(1), 55-84.
- Flannery, M.J., Bliss, R.R., 2019. Market discipline in regulation: Pre- and post-crisis in Berger, A.N., P. Molyneux, and J.O.S. Wilson (eds.) Oxford Handbook of Banking 3rd edition. Oxford: Oxford University Press.
- Fonseca, A.R., Gonzalez, F., 2008. Cross-country determinants of bank income smoothing my managing loan loss provisions. *Journal of Banking & Finance*, 32(2), 217-228.
- Francis, J., Isson, P.O., Schipper, K., 2006. Earnings quality. *Foundations and Trends in Accounting*, 1(4), 259-340.
- Gao, Y., Liao, S., Wang, X., 2018. Capital markets' assessment of the economic impact of the Dodd-Frank on systemically important financial firms. *Journal of Banking & Finance*, 86, 204-223.

- Garcia-Sanchez, I.M., Garcia-Meca, E., Cuadrado-Ballesteros, B., 2017. Do financial experts on audit committees matter for bank insolvency risk-taking? The monitoring role of bank regulation and ethical policy. *Journal of Business Research*, 76, 52-66.
- Greenawalt, M.B., Sinkey, J.F., 1988. Bank loan-loss provisions and the income-smoothing hypothesis: An empirical analysis, 1976-1984. *Journal of Financial Services Research*, 1, 301-318.
- Hagendorff, J., 2019. Corporate governance and culture in banking, in Berger, A.N., P. Molyneux, and J.O.S. Wilson (eds.) Oxford Handbook of Banking 3rd edition. Oxford: Oxford University Press.
- Hainmueller, J., 2012. Entropy Balancing for causal effects: A multivariate reweighting method to produce balanced samples in observational studies. *Political Analysis*, 20(1), 25-46.
- Healy, P.M., Palepu, K.G. 2003. The fall of Enron. *Journal of Economic Perspectives*, 17(2), 3-26.
- Hines, C.S., Masil, A., Mauldin, E.G., Peters, G.F., 2015. Board risk committees and audit pricing. *Auditing: A Journal of Practice & Theory*, 34(4), 59-84.
- Hoitash, U., Hoitash, R., Bédard, J.C., 2009. Corporate governance and internal control over financial reporting: A comparison of regulatory regimes. *Accounting Review*, 84(3), 839-867.
- Honig, S., 2009. Corporate governance changing rapidly with more federal control. *Michigan Lawyers Weekly* (August 10).
- Huizinga, H., Laeven, L., 2012. Bank valuation and accounting discretion during a financial crisis. *Journal of Financial Economics*, 106(3), 614–634.
- Huizinga, H., Laeven, L., 2019. The procyclicality of banking: Evidence from the euro area. *IMF Economic Review*, 67, 496-527.
- Ignatowski, M., Korte, J., 2014. Wishful thinking or effective threat? Tightening bank resolution regimes and bank risk-taking. 2014. *Journal of Financial Stability*, 15, 264-281.
- Ittonen, K., Tronnes, P., Vähämaa, S., 2018. Do former auditors on the audit committee constrain earnings management? Evidence from the banking industry. Available at SSRN: https://ssrn.com/abstract=2753710
- Jensen, M.C., Meckling, W.H., 1976. Theory of the firm: Managerial behavior, agency costs and ownership structure. *Journal of Financial Economics*, 3(4), 305-360.
- Jiang, L., Levine, R., Lin, C., 2016. Competition and bank opacity. *Review of Financial Studies*, 29(7), 1911-1942.
- Jones, J.S., Lee, W.Y., Yeager, T.H., 2012. Opaque banks, price discovery, and financial instability. *Journal of Financial Intermediation*, 21(3), 383-408.

- Kanagaretnam, K., Lobo, G.J., Mathieu, R., 2003. Managerial incentives for income smoothing through bank loan loss provisions. *Review of Quantitative Finance and Accounting*, 20(1), 63-80.
- Kanagaretnam, K., Lobo, G.J., Yang, D.H., 2004. Joint tests of signaling and income smoothing through bank loan loss provisions. *Contemporary Accounting Research*, 21(4), 843-884.
- Kanagaretnam, K., Lobo, G.J., Yang, D.H., 2005. Determinants of signaling by banks through loan loss provisions. *Journal of Business Research*, 58 (3), 312-320.
- Kanagaretnam, K., Krishnan, G., Lobo, G., 2009. Is the market valuation of banks' loan loss provision conditional on auditor reputation? *Journal of Banking & Finance*, 33(6), 1039-1047.
- Kanagaretnam, K., Lim, C.Y., Lobo, G.J., 2010. Auditor reputation and earnings management: International evidence. *Journal of Banking & Finance*, 34(10), 2318-2327.
- Karamanou, I., Vafeas, N., 2005. The association between corporate boards, audit committees, and management earnings forecasts: An empirical analysis. *Journal of Accounting Research*, 43(3), 453-486.
- Kim, M.S., Kross, W., 1998. The impact of the 1989 change in bank capital standards on loan loss provisions and loan write-offs. *Journal of Accounting and Economics*, 25(1), 69-99.
- Kirkpatrick, G., 2009. The Corporate Governance Lessons from the Financial Crisis. Paris: OECD.
- Klein, A., 2002. Audit committee, board of director characteristics and earnings management. *Journal of Accounting and Economics*, 33(3), 375-400.
- KPMG 2009. The Audit Committee Journey. New York: KPMG Audit Committee Institute.
- KPMG. 2014. Audit committee workload: Keeping an eye on the ball. London: KPMG
- Krishnan, J., 2005. Audit committee quality and internal control: An empirical analysis. *Accounting Review*, 80(2), 649–675.
- Krishnan, G.V., Visvanathan, G., 2008. Does the SOX definition of an accounting expert matter? The association between audit committee directors' accounting expertise and accounting conservatism. *Contemporary Accounting Research*, 25(3), 827-858.
- Krishnan, J., Wen, Y., Zhao, W., 2011. Legal expertise on corporate audit committees and financial reporting quality. *Accounting Review*, 86(6), 2099–2130.
- Laeven, L., Majnoni, G., 2003. Loan loss provisioning and economic slowdowns: Too much, too late? *Journal of Financial Intermediation*, 12(2), 178-197.
- LaFond and, R., You, H., 2010. The Federal Deposit Insurance Corporation Improvement Act, bank internal controls and financial reporting quality. *Journal of Accounting and Economics*, 49, 75-83.

- Leledakis, G.N., Pyrgiotakis, E.G., 2019. U.S bank M&As in the post-Dodd-Frank Act era: Do they create value? *Journal of Banking & Finance* (forthcoming).
- Lemmon, M., M.R., Roberts, 2010. The response of corporate financing and investment to changes in the supply of credit. *Journal of Financial and Quantitative Analysis* 45(3), 555-587.
- Lawson, B.P., Wang, D., 2016. The earnings quality information content of dividend policies and audit pricing. *Contemporary Accounting Research*, 33(4), 1685-1719.
- Linck, J.S., Netter, J.M., Yang, T., 2009. The effects and unintended consequences of the Sarbanes-Oxley Act on the supply and demand for directors. *Review of Financial Studies*, 22(8), 3287-3328.
- Liu, C.C, Ryan S.G., 1995. The effect of bank loan portfolio composition on the market reaction to and anticipation of loan loss provisions. *Journal of Accounting Research*, 33(1), 77-94.
- Liu, C.C., Ryan, S.G., 2006. Income smoothing over the business cycle: Changes in banks' coordinated management of provisions for loan losses and loan charge-offs from the pre-1990 bust to the 1990s boom. *Accounting Review*, 81(2), 421–441.
- Liu, H., Norden, L., Spargoli, F., 2020. Does uniqueness in banking matter? *Journal of Banking & Finance*, 120, 105941.
- Matsumoto, D., 2002. Management's incentives to avoid negative earnings surprises. *The Accounting Review*, 77(3), 483-514.
- Morgan, D.P., 2002. Rating banks: risk and uncertainty in an opaque industry. *American Economic Review*, 92(4), 874-888.
- Moyer, S.E., 1990. Capital adequacy ratio regulations and accounting choices in commercial banks. *Journal of Accounting and Economics*, 13(2), 123-154.
- McMullen, D.A., 1996. Audit committee performance: an investigation of the consequences associated with audit committee. *Auditing: A Journal of Theory and Practice*, 15(1), 87-103.
- National Association of Corporate Directors (NACD). 2000. *Report of the NACD Blue Ribbon Commission on Audit Committees: A Practical Guide.* Washington, DC: National Association of Corporate Directors.
- Nguyen, T., 2013. The disciplinary effect of subordinated debt on bank risk taking. *Journal of Empirical Finance* 23, pp.117-141.
- Norden, L., Stoian, A., 2014. Bank earnings management through loan loss provisions: A double-edged sword? *De Nederlandsche Bank Working Paper*, Number 404.
- Office of the Comptroller of the Currency (OCC). 2016. *The Director's Book: Role of Directors for National Banks and Federal Savings Associations.* Washington DC: OCC.

- Oster, E. 2019. Unobservable selection and coefficient stability: Theory and evidence. *Journal* of Business & Economic Statistics, 37(2), 187-204.
- Protiviti. 2011. Should the board have a separate risk committee? Board perspectives: Risk oversight. Available at:

http://www.protiviti.com/ja-JP/Downloads/RiskOversight_vol24_E.pdf

- Roberts, M.R., Whited, T., 2013. Endogeneity in empirical corporate finance. In: G.M. Constantinides, M. Harris, and R. M. Stulz (eds.) Handbook of the Economics of Finance. Amsterdam: Elsevier.
- Schipper, K., Vincent L., 2003. Earnings quality. *Accounting Horizons*, 17(Supp.), 97-110.
- Schmidt, J., Wilkins, M., 2013. Bringing darkness to light: The influence of auditor quality and audit committee expertise on the timeliness of financial statement restatement disclosures. *Auditing: A Journal of Practice and Theory*, 32(1), 221-244.
- Securities and Exchange Commission (SEC). 1999(b). *NYSE Rulemaking: Order Approving Proposed Rule Change Amending the Audit Committee Requirements and Notice of Filling and Order Granting Accelerated Approval of Amendments* No. 1 and No. 2. Washington, D.C.: SEC.
- Sharma, V., Naiker, V., Lee, B., 2009. Determinants of audit committee meeting frequency: evidence from a voluntary governance system. *Accounting Horizons*, 23(3), 245–263.
- Sharma, V.D., Iselin, E.R., 2012. The association between audit committee multipledirectorships, tenure, and financial misstatements. *Auditing: A Journal of Practice and Theory*, 31(3), 149–175.
- Shepardson, M.L., 2019. Effects of individual task-specific experience in audit committee oversight of financial reporting outcomes. *Accounting, Organizations and Society,* 74, 56-74.
- Shivdasani, A., Yermack, D., 1999. CEO involvement in the selection of new board members: An empirical analysis. *Journal of Finance*, 54(5),1829-1853.
- Sommer Jr., A.A., 1991. Auditing audit committees: An educational opportunity for auditors. *Accounting Horizons*, 5(2), 91-93.
- Srivastav, A., Hagendorff, J., 2016. Corporate governance and bank risk-taking. *Corporate Governance: An International Review*, 24(3), 334-345.
- Sun, J., Liu, J., 2014. Audit committees' oversight of bank risk-taking. *Journal of Banking & Finance*, 40, 376-387.
- Vafeas, N., 2005. Audit committees, boards, and the quality of reported earnings. *Contemporary Accounting Research*, 22(4), 1093-1122.
- Wall, L., Koch, T., 2000. Bank loan loss accounting: A review of theoretical and empirical evidence. *Economic Review*, Federal Reserve Bank of Atlanta, 85(2), 1-19.

- Wahlen, J., 1994. The nature of information in commercial bank loan loss disclosures. *The Accounting Review*, 69 (3), 455-478.
- Xie, B., Davidson, W.N., DaDalt, P.J., 2003. Earnings management and corporate governance: The roles of the board and the audit committee. *Journal of Corporate Finance*, 9(3), 295–316.
- Yang, J.S., Krishnan, J., 2005. Audit committees and quarterly earnings management. *International Journal of Auditing*, 9(3), 201-219.
- Zalewska, A., 2014. Challenges of corporate governance: Twenty years after Cadbury, ten years after Sarbanes–Oxley. *Journal of Empirical Finance* 27, 1-9.
- Zaman, M., 2001. Turnbull-generating undue expectations of the corporate governance role of audit committees. *Managerial Auditing Journal*, 16(1), 5-9.



Figure 1: Evolution of discretionary loan loss provisions from 2007 to 2016.

Note: This graph shows the evolution of discretionary loan loss provisions for affected and unaffected bank holding companies over the period 2007-2016. The dashed vertical line marks 2009, the year immediately before Section 165h of Dodd-Frank act came into effect.

Figure 2: Evolution of the discretionary loan loss provisions after the implementation of Section 165h of Dodd-Frank Act



Note: This graph shows the evolution of discretionary loan loss provisions for affected and unaffected bank holding companies over the period 2007-2016.

Variable	Definition
Panel A: Variables used in main analysis	
DLLP	The absolute value of the discretionary loan loss provisions (residuals from Eq. (1))
DLLP+	The positive value of the discretionary loan loss provisions (residuals from Eq. (1))
DLLP-	The negative value of the discretionary loan loss provisions (residuals from Eq. (1))
Post Event	Indicator variable equals 1 in the post- Dodd Frank Act period (2010), zero otherwise
Affected	Indicator variable equals 1 if bank holding companies have a joint audit and risk committee one year before the introduction of Section 165h (2009), zero
Audit Committee Size	The size of audit committee.
Financial Expert	The number of members in the audit committee who are financial experts.
Meetings	The number of audit committee meetings held by bank per year.
Intra-bank busyness	The percentage of directors of the audit committee who also sit on other committees in the same bank.
Inter-bank busyness	The percentage of directors of the audit committee who also sit on other committees in a different bank board.
Tenure	The average number of years the audit committee members serve as directors in the audit committee.
Size	The natural logarithm of total assets at the
Dividends	Total cash dividends paid to common shareholders divided by total equity
lag_LLP	One-year lag of loan loss provisions scaled
Capital	Bank capital as measured by Tier 1 capital
EBTLLP	Net income before taxes and loan loss provisions scaled by total assets.
ΔEBTLLP	One-year ahead change in net income before taxes and loan loss provisions scaled by total assets.

Table 1: Variable Definitions

Panel B: Definition of variables used to estimate discretionary loan loss provisions	
LLP	The ratio of loan loss provisions to beginning of the year total loans
ΔΝΡΑ	Change in non-performing assets over the year divided by beginning of the year total loans.
ΔLOANS	Change in total loans over the year divided by beginning of the year total loans.
CSRET	The return on the Case-Shiller Real Estate Index over the year.
ΔGSP	Change in GSP (gross state product) over the year scaled by 100.
ΔUNEMP	Change in unemployment rates over the year.
ALW	Loan loss allowances over the year divided by beginning of the year total loans.
CO	Net charge off over the year divided by beginning of the year total loans.
Panel C: Variables used in sensitivity analyses	
Joint Risk Committee	Indicator variable that equals one if a bank holding company has a joint risk with another committee in 2009 (one year prior to the Dodd-Frank Act), and zero otherwise
TARP	Binary variable that equals one for the period that a bank holding company receives and repays TARP funding, and zero otherwise
Tax	Binary variable that equals onethe year the state in which a bank holding company is headquartered inchanges its corporate income tax rate, and zero otherwise
Stress	Binary variable that equals one for the years a bank holding company is part of the stress tests, and zero otherwise.
Small Positive Earnings Changes	Indicator variable that equals one if earnings increase is positive and up to 1.5 percent over last year's net income divided by total assets, and zero otherwise.
Surprise Avoidance	Indicator variable that equals one if an earnings surprise is between 0.00 and 0.04 cents over the consensus (median) analyst forecast, measured as the last forecast prior to the announcement of annual earnings, and zero otherwise.

Note: This table provides the definitions of variables used in this study. *Panel A* shows the variables used in main analysis, *Panel B* shows the variables used to estimate discretionary loan loss provisions and *Panel C* shows the variables used in sensitivity analysis.

			11			Unaffected BHCs		
	Affec	ted BHCs	Unaffe	cted BHUS	(Entropy balancing)			
	N	Mean	Ν	Mean	Difference	Ν	Mean	Difference
	(1)	(2)	(3)	(4)	(5)=(2)-(4)	(6)	(7)	(8)=(2)-(7)
Panel A: Pre-treatmen	nt peri	od (2007-2	009)					
DLLP	54	0.0047	63	0.0029	0.0018	63	0.0023	0.0024
DLLP+	19	0.0067	38	0.0038	0.0029	38	0.0031	0.0036
DLLP-	35	-0.0036	25	-0.0015	-0.0021	25	-0.0012	-0.0024
Audit Committee Size	54	5	63	5	0.0000	63	5	0.0000
Financial Expert	54	2	63	3	-1.0000	63	3	-1.0000
Meetings	54	10	63	11	-1.0000	63	11	-1.0000
Intra-bank busyness	54	0.8403	63	0.0232	0.8170	63	0.0358	0.8045
Inter-bank busyness	54	0.2627	63	0.3585	-0.0958	63	0.2259	0.0368
Tenure	54	4	63	4	0.0000	63	4	0.0000
Size	54	17.2962	63	18.2512	-0.9549	63	17.3152	-0.0019
Dividends	54	0.0390	63	0.0365	0.0025	63	0.0298	0.0092
lag_LLP	54	0.0093	63	0.0095	-0.0002	63	0.0077	0.0016
Capital	54	0.1146	63	0.1094	0.0052	63	0.1128	0.0018
EBTLLP	54	0.0173	63	0.0150	0.0023	63	0.0149	0.0024
ΔEBTLLP	54	0.0001	63	0.0003	-0.0002	63	0.0006	-0.0005
Panel B: Post-treatme	ent per	iod (2010-	2016)					
DLLP	152	0.0024	153	0.0020	0.0004	153	0.0017	0.0007
DLLP+	82	0.0021	79	0.0014	0.0007	79	0.0015	0.0006
DLLP-	70	-0.0028	74	-0.0026	-0.0002	74	-0.0021	-0.0007
Audit Committee Size	152	5	153	6	-1.0000	153	5	0.0000
Financial Expert	152	3	153	3	0.0000	153	3	0.0000
Meetings	152	10	153	11	-1.0000	153	10	-0.0000
Intra-bank busyness	152	0.2972	153	0.0464	0.2508	153	0.0804	0.2168
Inter-bank busyness	152	0.2390	153	0.4385	-0.1995		0.3496	-0.1106
Tenure	152	6	153	5	1.0000	153	5	1.0000
Size	152	17.8683	153	18.4422	-0.5739	153	17.6194	0.2489
Dividends	152	0.0247	153	0.0186	0.0061	153	0.0178	0.0069
lag_LLP	152	0.0101	153	0.0101	0.0000	153	0.0103	-0.0002
Capital	152	0.1339	153	0.1270	0.0069	153	0.1260	0.0079
EBTLLP	152	0.0189	153	0.0148	0.0041	153	0.0150	0.0039
ΔEBTLLP	152	-0.0002	153	-0.0003	0.0001	153	-0.0007	0.0005
Panel C: Pre-trend gro	owth ra	ate in discr	etionary	y loan loss p	provisions (200)7-2009)		
2007	18	2.0084	21	1.6242	0.3842	21	1.6855	0.3229
2008	18	2.4128	21	36.3507	-33.9379	21	2.0715	0.3413
2009	18	1.2989	21	2.2979	-0.9990	21	1.1131	0.1858

Table 2: Summary Statistics

Note: This table reports summary statistics for our sample. Panel A and B show summary statistics for the treated and control group before and after the entropy balancing procedure, for both the pre-treatment period (2007-2009) and the post-treatment period (2010-2016), respectively. Panel C presents trends in the pre-treatment period and the mean comparison of these trends between treated and control banks for the outcome variable. The definitions of the variables are given in Table 1.

	DLLP	DLLP+	DLLP-
	(1)	(2)	(3)
Affected x Post Event	-0.0022***	-0.0028***	0.0014
	(0.0006)	(0.0009)	(0.0011)
Audit Committee Size	0.0001	0.0003	0.0000
	(0.0001)	(0.0001)	(0.0002)
Financial Expert	-0.0000	-0.0001	0.0002
	(0.0001)	(0.0001)	(0.0002)
Meetings	0.0001**	0.0000	-0.0000
	(0.0000)	(0.0000)	(0.0000)
Intra-bank busyness	-0.0002	-0.0000	0.0005
	(0.0006)	(0.0008)	(0.0006)
Inter-bank busyness	-0.0000	-0.0000	0.0004
	(0.0008)	(0.0001)	(0.0024)
Tenure	0.0002*	-0.0001	-0.0005**
	(0.0001)	(0.0001)	(0.0002)
Size	-0.0007	0.0020	0.0051***
	(0.0011)	(0.0013)	(0.0013)
Dividends	0.0079	-0.0125	-0.0141
	(0.0130)	(0.0238)	(0.0193)
lag_LLP	0.0273	0.0373	-0.0341
	(0.0312)	(0.0295)	(0.0407)
Capital	-0.0235**	-0.0121	0.0333
	(0.0099)	(0.0121)	(0.0237)
EBTLLP	-0.0546	-0.0403	0.0865
	(0.0398)	(0.0458)	(0.0831)
ΔEBTLLP	-0.0563**	-0.0248	0.1652**
	(0.0230)	(0.0277)	(0.0677)
Bank fixed effects	Yes	Yes	Yes
Year fixed effects	Yes	Yes	Yes
Observations	422	218	204
No. of banks	45	45	45
R-squared	0.274	0.490	0.348

 Table 3: Audit Committees and Bank Financial Reporting Quality-Baseline Results

Note: This table reports the baseline results. The dependent variable is the absolute value of discretionary loan loss provisions. Column 1 investigates the effect of the separation of audit and risk committees following the introduction of section 165h of the Dodd-Frank Act. The variable of interest is *Affected x Post Event* which indicates the improvement in financial reporting quality between affected and unaffected banks following the introduction of section 165h of the Dodd-Frank Act. Column 2 uses positive discretionary loan loss provisions to investigate the impact of the separation of audit and risk committees on bank financial reporting quality. Column 3 uses negative discretionary loan loss provisions to investigate the impact of the separation of audit and risk committees on bank financial reporting quality. All regressions include the following set of control variables: Audit Committee Size, Financial Expert Meetings, Intra-bank busyness, Inter-bank busyness, Tenure, Size, Dividends, Loan loss provisions (lagged by one period), Capital, Earnings before taxes and loan loss provisions. The definitions of variables are provided in Table 1. Standard errors clustered at bank level are reported in parentheses. ***, **, * indicate significance at the 1%, 5% and 10% level, respectively.

	Size	Financial	Meetings	Intra-bank	Inter-bank	Tenure	Credit risk
		Expert		busyness	busyness		
	(1)	(2)	(3)	(4)	(5)	(6)	(7)
Affected x Post Event	0.0165	0.1054	0.0880	-0.5495***	-0.1769***	0.1777	-0.0000
	(0.0764)	(0.1322)	(0.1024)	(0.0898)	(0.0414)	(0.1121)	(0.0000)
Bank level controls	Yes	Yes	Yes	Yes	Yes	Yes	Yes
A.C. Characteristics	No	No	No	No	No	No	Yes
Bank fixed Effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Year fixed Effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Observations	422	422	422	422	422	422	422
No. of banks	45	45	45	45	45	45	45
R-squared	0.032	0.101	0.125	0.308	0.104	0277	0.122

Table 4: Audit Committees and Bank Financial Reporting Quality-Possible Channels

Note: This table reports the results of the mechanism analyses (columns 1-6), as well as that of testing for an alternative explanation (column 7) for our observed baseline results. Columns 1-7 report the estimates of Equation (2) for different dependent variables. In column 1 the dependent variable, Size is defined as the natural logarithm of the size of the audit committee. In column 2 the dependent variable, Financial Expert is defined as the natural logarithm of the number of audit committee members that are considered as financial experts. In column 3 the dependent variable, Meetings is defined as the natural logarithm of meetings held by audit committees. In column 4 the dependent variable, Intra-bank busyness is defined as the percentage of audit committee members that sit also in other committees in the same bank. In column 5 the dependent variable, Inter-bank busyness is defined as the percentage of audit committee members that sit also in other committees in a different bank. In column 6 the dependent variable, *Tenure* is defined as the natural logarithm value of the years that directors serve on the audit committee. In column 7 the dependent variable, Credit risk is defined as the ratio of non-performing loans to total loans. All regressions include the following set of control variables but are not reported in the table for brevity: Size, Dividends Loan loss provisions (lagged by one period), Capital, Earnings before taxes and loan loss provisions, One-year forward change in earnings before taxes and loan loss provisions. Audit committee characteristics including: Audit Committee Size, Financial Expert Meetings, Intra-bank busyness, Inter-bank busyness, Tenure are only included in the specification presented in column 7. The definitions of variables are provided in Table 1. Standard errors clustered at bank level are reported in parentheses. ***, **, * indicate significance at the 1%, 5% and 10% level, respectively.

Table 5: Robustness Tests

	Placebo	Covariates	Two-period
	Event	Exclusion	Sample
	(1)	(2)	(3)
Affected x Placebo Post Event	0.0011		
	(0.0011)		
Affected x Post Event		-0.0015***	-0.0017***
		(0.0003)	(0.0002)
Bank level controls	Yes	No	Yes
Bank fixed Effects	Yes	Yes	Yes
Year fixed Effects	Yes	Yes	No
Observations	292	422	84
No. of banks	39	45	45
R-squared	0.358	0.231	0.664

Note: This table presents the results of sensitivity checks of our baseline regressions with respect to different model specifications and sample composition, as well as on the validity of the parallel trend assumption. In column (1), we create a hypothetical event two years prior the actual year of event in 2008. The results are estimated using a sample spanning the period before the introduction of Section 165h of Dodd-Frank Act. *Placebo Post Event* is a dummy variable equal to ones for years 2007-2009 and zero for years 2002-2006. In column (2), we exclude covariates from the main model. In column (3), following Bertrand et al. (2004), we collapse our dataset into a two-period panel. Specifically, we average the data before (2007-2009) and after (2011-2016) the separation of audit and risk committees by the introduction of Section 165h of the Dodd-Frank Act in 2010. Column (1) and (3) include a set of control variables: Audit Committee Size, Financial Expert Meetings, Intra-bank busyness, Inter-bank busyness, Tenure, Size, Dividends Loan loss provisions (lagged by one period), Capital, Earnings before taxes and loan loss provisions, one year forward change in earnings before taxes and loan loss provisions but are not reported in the table for brevity. Standard errors, clustered at bank level, are reported in parentheses. ***, **, * indicate significance at the 1%, 5% and 10% level, respectively. The definitions and sources of variables are given in Table 1.

Table 6: Further Robustness Tests

	Joint Risk	Expanded	Split
	Committee	Sample	Committee
	(1)	(2)	(3)
Joint Risk Committee x Post Event	-0.0003		
	(0.0006)		
		-0.0011**	-0.0024***
Affected x Post Event		(0.0005)	(0.0006)
Bank level controls	Yes	Yes	Yes
Bank fixed Effects	Yes	Yes	Yes
Year fixed Effects	Yes	Yes	Yes
Observations	280	499	422
No. of banks	30	53	45
R-squared	0.159	0.239	0.263

Note: This table presents the results of sensitivity checks of our baseline regressions with respect to different model specifications and sample composition. In column (1), the model is estimated using as a treated group bank holding companies that have a joint risk with another committee (e.g., compliance committee, capital committee, examination committee, etc.) and as a control group the banks that had already an independent risk committee before the introduction of Section 165h. In column (2), we re-estimate the baseline model using an expanded sample that also includes banks that grew in size to exceed the regulatory threshold of \$10 billion as of June 2014. In column (3), we re-estimate the baseline model by changing the treatment period to start after the separation of audit and risk committee has occurred for each bank holding company. In all regressions, we include a set of control variables: Audit Committee Size, Financial Expert Meetings, Intra-bank busyness, Inter-bank busyness, Tenure, Size, Dividends Loan loss provisions (lagged by one period), Capital, Earnings before taxes and loan loss provisions, one-year forward change in earnings before taxes and loan loss provisions but are not reported in the table for brevity. Standard errors, clustered at bank level, are reported in parentheses. ***, **, * indicate significance at the 1%, 5% and 10% level, respectively. The definitions and sources of variables are given in Table 1.

Table 7: Confounding Events

	TARP	State Corporate Income Tax	Stress Test
	(1)	(2)	(3)
Affected x Post Event	-0.0017***	-0.0023***	-0.0022***
	(0.0004)	(0.0006)	(0.0007)
TARP	-0.0006		
	(0.0011)		
Affected x TARP	0.0014		
	(0.0010)		
Tax		-0.0015***	
		(0.0005)	
Affected x Tax		0.0037**	
		(0.0014)	
Stress			-0.0007
			(0.0005)
Affected x Stress			-0.0015
			(0.0015)
Bank level controls	Yes	Yes	Yes
Bank fixed Effects	Yes	Yes	Yes
Year fixed Effects	Yes	Yes	Yes
Observations	422	422	422
No. of banks	45	45	45
R-squared	0.280	0.284	0.291

Notes: This table presents results of the effect of the separation of audit and risk committees following the introduction of section 165h of the Dodd-Frank Act, on bank financial reporting quality, while considering potential confounding events. The dependent variable is discretionary loan loss provisions. The variable of interest is Affected x Post Event which indicates the improvement in financial reporting quality between affected and unaffected banks following the introduction of section 165h of the Dodd-Frank Act. In column (1), we include an additional interaction term between the dummy for TARP and variable Affected. TARP is defined as a binary variable which equals 1 for the period that bank holding companies purchase and repay the funding. Column (2) includes an additional interaction term between the dummy for Tax and variable Affected. Tax is defined as a binary variable which equals 1 for the year of our sample each state has changed corporate income tax rate relative to previous and next year. Column (3) includes an additional interaction term between the dummy for Stress and variable Affected. Stress is defined as a binary variable which equals 1 for the years that bank holding companies are part of the stress tests. The definitions of variables are given in Table 1. All model specifications are estimated using OLS. In all regressions, we include a set of control variables: In all regressions, we include a set of control variables: Audit Committee Size, Financial Expert Meetings, Intra-bank busyness, Inter-bank busyness, Tenure, Size, Dividends Loan loss provisions (lagged by one period), Capital, Earnings before taxes and loan loss provisions, one-year forward change in earnings before taxes and loan loss provisions but are not reported in the table for brevity. Standard errors, clustered at bank level, are reported in parentheses. ***, **, * indicate significance at the 1%, 5% and 10% level, respectively.

	Small positive	Surprise	
	earnings changes	avoidance	
	(1)	(2)	
Affected x Post Event	-0.0540*	-0.1772*	
	(0.0290)	(0.0948)	
Bank level controls	Yes	Yes	
Bank fixed Effects	Yes	Yes	
Year fixed Effects	Yes	Yes	
Observations	477	477	
No. of banks	45	45	
R-squared	0.041	0.166	

Table 8: Alternative Measures of Financial Reporting Quality

Note: This table reports the results of sensitivity checks of our baseline estimation with respect to alternative measures of financial reporting quality. In column (1), the dependent variable is replaced with the variable small positive earnings changes. In column (2), the dependent variable is replaced with the variable surprise avoidance. In all regressions, we include a set of control variables: Audit Committee Size, Financial Expert Meetings, Intrabank busyness, Inter-bank busyness, Tenure, Size, Dividends Loan loss provisions (lagged by one period), Capital, Earnings before taxes and loan loss provisions, one-year forward change in earnings before taxes and loan loss provisions but are not reported in the table for brevity. Standard errors, clustered at bank level, are reported in parentheses. ***, **, * indicate significance at the 1%, 5% and 10% level, respectively.

Table A1: Summary Statistics for the variables used in Equation (1)

			(1)				(2)			(3)	
			All BHCs			Affe	cted BHCs			Unaffecte	ed BHCs	
	Ν	Mean	Median	Std	Ν	Mean	Median	Std	Ν	Mean	Median	Std
LLP	422	0.0098	0.0046	0.0120	206	0.0097	0.0047	0.0120	216	0.0099	0.0044	0.0121
ΔNPA	422	0.0015	-0.0005	0.0177	206	0.0020	-0.0002	0.0206	216	0.0010	-0.0009	0.0145
ΔLOANS	422	0.0684	0.0481	0.1586	206	0.0755	0.0574	0.1383	216	0.0617	0.0386	0.1759
ALW	422	0.0178	0.0153	0.0100	206	0.0178	0.0149	0.0106	216	0.0178	0.0157	0.0094
CO	422	-0.0096	-0.0050	0.0111	206	-0.0095	-0.0039	0.0120	216	-0.0096	-0.0056	0.0102
ΔGSP	422	3.7424	5.0300	11.6158	206	3.9097	4.8500	12.1581	216	3.5821	5.6650	11.0972
ΔUNEMP	422	-0.0215	-0.4000	1.3811	206	-0.0608	-0.4000	1.3500	216	0.0162	-0.3000	1.4123
CSRET	422	375.450	325.770	136.089	206	386.971	335.630	144.241	216	364.409	322.770	127.146
No. of banks	45				23				22			

Note: This table presents the summary statistics used in Equation 1. Column (1) reports statistics for the general sample, while column (2) and (3) report statistics by treatment status. The definitions of the variables are given in Table 1.

	(1)
	Loan loss provisions
ΔNPA_{it+1}	0.0969*
	(0.0489)
ΔNPA_{it}	0.0288**
	(0.0120)
ΔNPA_{it-1}	0.0397*
	(0.0215)
$SIZE_{it-1}$	0.0000
	(0.0001)
$\Delta LOANS_{it}$	-0.0004
t t	(0.0019)
ALW_{it-1}	0.1718***
	(0.0526)
CO _{it}	-0.7247***
	(0.0624)
<i>CSRET</i> _{it}	0.0000
	(0.0000)
ΔGSP_{it}	-0.0000**
	(0.0000)
$\Delta UNEMP_{it}$	0.0014***
	(0.0003)
Observations	422
R-squared	0.8822

 Table A2: Stage-one regression for estimating discretionary loan loss provisions

Notes: This table reports the results from the estimation of Equation 1 using OLS. The sample consists of 45 BHCs from 2007 through 2016. Standard errors clustered at bank level are reported in parentheses. ***, **, * indicate significance at the 1%, 5% and 10% level, respectively. The definitions of variables are given in Table 1.



Recent CRBF Working papers published in this Series

Fourth Quarter | 2021

21-014 Giovanni Cerulli, Franco Fiordelisi, David Marques-Ibanez: The Burden of Bank Supervision.

21-013 **Simon Fritzsch, Maike Timphus, Gregor Weiß:** Marginals versus Copulas: Which Account for More Model Risk in Multivariate Risk Forecasting?

Third Quarter | 2021

21-012 Simon Fritzsch, Felix Irresberger, Gregor Weiß: Cross-Section of Option Returns and the Volatility Risk Premium.

21-011 Nhan Le, Duc Duy Nguyen, Vathunyoo Sila: Does Shareholder Litigation Affect the Corporate Information Environment?

21-010 **Hiep N. Luu, Linh H. Nguyen, John O.S. Wilson:** Organizational Culture, Competition and Bank Loan Loss Provisions.

21-009 Alessio Reghezza, Yener Altunbas, David Marques-Ibanez, Costanza Rodriguez d'Acri, Martina Spaggiari: Do Banks Fuel Climate Change?

Second Quarter | 2021

21-008 **Dolores Gutiérrez-Mora and Daniel Oto-Peralías:** Gendered Cities: Studying Urban Gender Bias through Street Names.

21-007 Linh H. Nguyen, John O.S. Wilson, Tuan Q. Le, Hiep N. Luu, Vinh X. Vo, Tram-Anh Nguyen: Deposit Insurance and Credit Union Lending.

21-006 Michael Koetter, Thomas Krause, Eleonora Sfrappini, Lena Tonzer: Completing the European Banking Union: Capital Cost Consequences for Credit Providers and Corporate Borrowers.



600 YEARS 1413 – 2013

University of St Andrews

Scotland's first university

